Supply Response to Consumer Inertia: Strategic Pricing in Medicare Part D

by

Yufei Wu

B.S. Economics and Finance, Tsinghua University (2011)

Submitted to the Department of Economics in partial fulfillment of the requirements for the degree of

Master of Science in Economics

at the

ADCHIVES							
LIBRARIES							
JUN 102016							
OF TECHNOLOGY							
OF TECHNOLOGY							

Contraction of the local division of the loc

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

June 2016

Author

Certified by

Accepted by

O'Yufei Wu All rights reserved. The author hereby grants to MIT permission to reproduce and to distribute publicly paper and electronic copies of this thesis document in whole or in part in any medium now known or hereafter created.

Signature redacted

Department of Economics March 1, 2016

Signature redacted

Amy Finkelstein Ford Professor of Economics Thesis Supervisor

Signature redacted

Ricardo Caballero Ford International Professor of Economics Chairman, Departmental Committee on Graduate Studies

Supply Response to Consumer Inertia: Strategic Pricing in Medicare Part D

by

Yufei Wu

Submitted to the Department of Economics on March 1, 2016, in partial fulfillment of the requirements for the degree of Master of Science in Economics

Abstract

A growing literature has documented evidence that consumers in health insurance markets are inertial, or behave as though they face substantial switching costs in choosing a health insurance plan. I investigate whether the private firms that provide prescription drug insurance through Medicare Part D exploit this inertia when setting prices. I first document descriptive evidence consistent with insurers initially setting low prices in order to "invest" in future demand before later raising prices to "harvest" inertial consumers. I then apply a two-step estimation approach following Bajari, Benkard and Levin (2007) to explore the implications of these invest and harvest incentives for equilibrium pricing, finding that on net, demand inertia reduces equilibrium prices (i.e. the invest incentive dominates the harvest incentive). Finally, I evaluate welfare consequences of policies that could be used to constrain insurers' ability to conduct such "invest-then-harvest" pricing patterns. I find, for example, that a policy change to cap premium increases would improve consumer welfare by both lowering average premiums and smoothing prices over time.

Thesis Supervisor: Amy Finkelstein Title: Ford Professor of Economics

Contents

Introd	uction \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 11
Empir	ical Setting and Data
0.2.1	Institutional Features of Medicare Part D
0.2.2	Data
Conce	ptual Framework
Descri	ptive evidence
0.4.1	Switching Costs
0.4.2	Invest-then-Harvest Pricing
0.4.3	Addressing Alternative Explanations
Model	
0.5.1	Demand
0.5.2	Cost
0.5.3	Supply
	0.5.3.1 Value Function
	0.5.3.2 Annual Profit Function
Struct	ural Estimation
0.6.1	Demand Estimation
0.6.2	Cost Estimation
0.6.3	Supply Estimation
	0.6.3.1 Step One: Empirical Pricing Policy Function
	0.6.3.2 Step Two: Uncover Insurers' Discount Factor
Count	er-factual Analysis
0.7.1	Do Switching Costs Lead to Higher or Lower Prices?
0.7.2	Policy Experiments
073	Implementation
0.1.0	-
0.7.4	Welfare Metrics
0.7.4 0.7.5	Welfare Metrics 37 Policy Experiment Results 38
0.7.4 0.7.5 Conclu	Welfare Metrics 37 Policy Experiment Results 38 usion 39
	Empir: 0.2.1 0.2.2 Concep Descrip 0.4.1 0.4.2 0.4.3 Model 0.5.1 0.5.2 0.5.3 Struct 0.6.1 0.6.2 0.6.3 Count 0.7.1 0.7.2

Bibliography

Acknowledgments

I am greatly indebted to my advisors Amy Finkelstein, Heidi Williams and Paulo Somaini for their invaluable guidance and support throughout this project. I thank Nikhil Agarwal, Jie Bai, Joseph Doyle, Esther Duflo, Glenn Ellison, Sara Fisher Ellison, Jonathan Gruber, Gaston Illanes, Yusuke Narita, Manisha Padi, Ariel Pakes, Maria Polyakova, Jim Poterba, Brendan Price, Michael Whinston, Hongkai Zhang and participants at the MIT Public Finance and Industrial Organization field lunches and workshops for very helpful comments, suggestions and discussions. I also thank Mohan Ramanujan and Jean Roth at NBER, and Sarah Brunsberg and Kelly McMaken at ResDAC for their help with navigating the Medicare data. I gratefully acknowledge the generous support from the MIT George and Obie Shultz Fund.

List of Tables

.

1	Summary Statistics	42
2	Summary Statistics on Plans by Year	43
3	Enrollment Shares as of 2011 by Beneficiary Cohort	43
4	Comparing Markups Between Entrant Plans and Incumbent Plans	44
5	Comparing Markups Between Entrant Plans and Incumbent Plans on Subsamples $\ .$	45
6	Polyakova (2015)'s Demand Estimates for Standard Enrollees	46
7	Empirical Pricing Strategies	47
8	Structural Parameter Estimate	47
9	Decomposition Results: Equilibrium Markup Levels	48
10	Counter-factual Policy Experiments: Equilibrium Markup Levels	48
11	Counter-factual Policy Experiments: Consumer Welfare	49
12	Counter-factual Policy Experiments: Social Welfare	49
A.1	Individual Cost Estimation	50

List of Figures

0-1	Markup By Year of Entry	40
0-2	Expected Claims Cost, Switching Cost and Preference by Consumer Type	41
A .1	Expected claims cost, switching cost and preference by consumer type $\ldots \ldots \ldots$	50

0.1 Introduction

A growing literature has documented evidence that consumers in health insurance markets behave as if they face substantial switching costs when choosing health insurance plans. In this paper, I investigate whether private firms exploit this type of consumer inertia when setting prices for health insurance products and analyze the resulting welfare and policy implications. My empirical setting is Medicare Part D, a public program in which private insurers under contract with the government provide outpatient prescription drug insurance to more than 30 million Medicare beneficiaries (Hoadley et al., 2014).

Consumer inertia is a well-recognized feature of Medicare Part D, where standard enrollees only need to actively choose plans when they first join the program and are subsequently defaulted into previous choices unless they choose to switch. Hoadley et al. (2012) document the low frequency of switching despite large changes in plan premiums. Miller and Yeo (2015a) and Polyakova (2015) both identify substantial switching costs among Part D enrollees and estimate significant welfare loss because switching costs tend to prevent consumers from re-optimizing and to lock them into sub-optimal plans.

Building on prior evidence of inertia in consumer demand in the context of Medicare Part D, my paper proceeds in three steps. First, I use administrative micro-data on Medicare beneficiaries and their plan choices to document descriptive facts that are consistent with the theoretical framework outlined by Klemperer (1987). The key idea of Klemperer (1987)'s framework is that, in the presence of demand inertia, insurers initially set low prices in order to "invest" in future demand before raising prices to "harvest" inertial consumers. I start by testing for this "invest-then-harvest" pricing pattern (also known as "bait-and-switch" or "bargains-then-ripoffs" pricing) by using a measure of markup or variable profit margin to eliminate potential confounding variations in cost and subsidy that affect insurers' pricing decisions. I document descriptive evidence consistent with insurers initially setting low prices in order to invest in future demand before later raising prices to harvest inertial consumers. Indeed, insurers charge lower markups when plans first enter and increase markups afterward; even the same insurer charges 148 dollars lower in annual markups for entrant plans than incumbent plans offering similar coverage. This difference in markup is over 30 percent of the average annual premium during the sample period. Such a striking invest-then-harvest pricing pattern rejects the null of no strategic response and provides suggestive evidence that insurers account for inertia when setting prices.

My finding confirms the public suspicion of Part D sponsors' "bait-and-switch" tactic. According to a Boston Globe article by Krasner (2006), the start-up of Medicare Part D was seen as "a once-ina-lifetime opportunity" to attract new customers for Humana, one of the biggest insurers operating in Medicare Part D. The article notes that Humana tends to introduce plans at low prices, which are subsequently increased by a large margin.¹ One health-care analyst's response to Humana's pricing is very telling: "That's not an acceptable inflationary increase in prices. That's sucker them in and you just start raising the prices." A Humana spokesman blamed the price increase on the government's subsidy formula, but that contention was disputed by an actuary from the Centers for Medicare and Medicaid Services (Krasner, 2006).

Insurers' invest-then-harvest pricing has important welfare implications. On one hand, dynamic choice inefficiency arises as consumers' plan choices tend not to remain optimal after price changes, but switching frictions prevent many from taking advantage of re-optimization. On the other hand, it is an empirical question whether the invest or harvest incentive dominates and whether prices are higher or lower compared with the benchmark with no inertia. To explore the implications of these invest and harvest incentives for equilibrium pricing, I propose and estimate a dynamic model of insurers' pricing decisions that incorporates consumer inertia and adverse selection. Following Bajari et al. (2007)'s two-step estimation approach, I uncover insurers' discount factor, which tells us how much firms value future profits relative to current profits and quantifies their incentive to invest in future demand. As a result, the identification comes from the observed price or markup levels. Intuitively, the more insurers care about the future, the stronger invest incentive they face and the lower they set their premiums. The structural estimation reveals a strong invest incentive for insurers, which is consistent with low markups observed early on.

I apply this dynamic model to answer two important economic questions. First, what is the net effect of strategic pricing in response to inertia on equilibrium prices? In other words, do switching costs toughen or soften competition? It is an empirical question and depends on which of the following incentive dominates – the incentive to price low to invest in future demand, or the incentive to harvest inertial incumbent consumers. To quantify insurers' trade-off between these counteracting incentives, I decompose observed pricing patterns by comparing this dynamic model with a counter-factual benchmark without inertia and with a counter-factual in which insurers are myopic and do not invest in future demand. Comparisons show that on net, demand inertia toughens competition and reduces equilibrium prices in this setting, i.e. the invest incentive dominates the harvest incentive.

¹For example, premiums of Humana Standard, with over 2 million enrollees, increased by 60 percent on average between 2006 and 2007 and by 466 percent in seven regions (Krasner, 2006).

Finally, I apply the model to understand the potential role of government regulations. What are the price and welfare consequences of policies that could be used to constrain insurers' ability to exploit inertia using the "invest-then-harvest" pricing tactic? Even if there are policies that can effectively reduce the scope of investing and harvesting, the effects of government intervention are not directly intuitive. In fact, the effects are ambiguous ex-ante because pricing response to inertia creates two offsetting effects on consumer welfare. On one hand, price increases create dynamic choice inefficiency in consumer choice in the presence of switching frictions. On the other hand, the structural estimation suggests that inertia reduces prices as insurers face very strong incentives to invest in future profits. The desirability of government intervention depends on how effectively each policy can smooth prices over time without increasing average price levels. In order to assess desirability of government intervention, I first consider the most natural and straightforward policy, which is to cap the rate of annual premium increases at ten percent.² A second policy I consider is to offer a public option at a low price to compete with private insurers. An inexpensive public option would not only restrain room for increasing prices later on, but would also reduce the incentive to invest in future demand early on. I also consider the effects of removing risk sharing and fully exposing insurers to excessive losses and gains from their pricing decisions. A caveat with the last two policies is that public options and risk corridors are important policy instruments with many potential effects other than influencing insurer response to inertia, and my analysis here only speaks to one of many aspects of their effects. Among these policies, I find that a policy change to cap premium increases would be the most effective in improving consumer welfare by both lowering average premiums and smoothing prices over time. Offering a low-price public option lowers average prices and increases consumer welfare, but such welfare gains are dominated by the extra social cost of offering the public plan. Removing risk sharing has little impact on welfare but transfers money from the government to insurers because with risk sharing, taxes on excessive gains outweigh subsidies on excessive losses, both in the model and empirically.

My work builds on multiple literatures and contributes to the general understanding of supply in privatized health insurance markets, often with switching frictions. In recent years, we have seen a growing role for non-group insurance over typical employer-based and traditional governmentprovided insurance. For example, the Affordable Care Act establishes state-based health insurance exchanges where individuals and small business can choose from plans provided by private insurers. Therefore, it is increasingly important to understand how the private supply side operates in health insurance markets with switching frictions. First, my paper builds on the growing literature on consumer inertia and choice frictions in general³ in insurance markets, including Medicare

²This policy experiment is similar in nature to the "Effective Rate Review" policy under the Affordable Care Act, which ensures that "in any State, any proposed rate increase by individual or small group market insurers at or above 10 percent will be scrutinized by independent experts to make sure it is justified". See CMS report: http://www.cms.gov/CCIIO/Resources/Fact-Sheets-and-FAQs/rate_review_fact_sheet.html.

³This growing body of literature examines inefficiency or sub-optimality of enrollees' plan choices in Medicare Part D (Heiss et al., 2008, 2013; Abaluck and Gruber, 2011, 2013; Kesternich et al., 2013; Kling et al., 2012; Ketcham et al., 2012, 2015).

Part D and other health insurance settings. Polyakova (2015) models inertial consumers facing a switching cost and estimates switching costs to be two to four times as high as annual premiums among Medicare Part D enrollees. Ho et al. (2015) study inattention as a crucial driver of observed inertia and analyze its implications for prices, consumer out-of-pocket costs and government subsidy. Switching friction is a general feature of a variety of insurance markets with defaults, not limited to prescription drugs for the elderly. For example, Nosal (2012) estimates switching cost in Medicare Advantage, while Handel (2014) provides evidence of consumer inertia among a large firm's employees in choosing from employer-provided insurance plans. The contribution of my paper is to build on these studies and develop a structural model of dynamic pricing that allows me to simulate supply-side policy counter-factuals.

Furthermore, my work contributes to recent studies on insurance supply in privatized health insurance markets and its interactions with government regulations. Abaluck and Gruber (2013) conclude that the increased welfare loss from choice inconsistency in Medicare Part D is largely driven by supply-side changes, indicating the importance of understanding insurers' behavior. Ericson (2014) is the first to examine strategic pricing in response to inertia in Medicare Part D, documenting evidence of increasing premiums that is consistent with insurers exploiting inertia in pricing. Miller (2014) studies the role of inertia as well as government subsidy in insurers' plan offering and welfare in Medicare Advantage. Shepard (2015) studies insurers' competition over hospital networks in response to adverse selection. Starc (2014) analyzes the impact of imperfect competition on consumer welfare in Medigap. Decarolis (2015) identifies insurers' strategic response to the low-income subsidy system in their plan offering and pricing. Decarolis et al. (2015) and Miller (2015) study the welfare impacts of the current subsidy policy in Medicare Part D. Ericson and Starc (2015) examine the impacts of pricing regulations in Massachusetts's health insurance exchange. Miller and Yeo (2015b) analyze the effect of introducing a public option alongside private insurers in Medicare Part D. Building on these papers, my study investigates insurers' pricing response to inertia and analyzes policy counter-factuals where such strategic pricing interacts with pricing regulation, additional competition from a public option, etc.

Finally, this study is related to both theory and empirical literatures on firm strategy in the presence of switching costs (Farrell and Klemperer, 2007, provide a review). Klemperer (1987) uses a two-period model to discuss the general intuition for firms' pricing incentives when consumers face switching cost – the incentive to invest in future demand by charging low prices, and the incentive to harvest inertial incumbent enrollees by charging high prices. Beggs and Klemperer (1992) illustrate how these two incentives interact in an infinite-period model with horizontal differentiation and infinite switching costs, and show analytically that the harvest incentive always dominates and switching costs soften competition. My study builds on Dubé et al. (2009) and Arie and Grieco (2014), both of which relax the crucial assumption of infinite switching cost and show that switching costs do not necessarily soften competition and can actually reduce equilibrium prices. Empirical evidence of strategic pricing in response to inertia is established in the bank deposit market (Sharpe, 1997), in the credit card market (Stango, 2002), in electricity markets (Waterson, 2003), in phone

services (Knittel, 1997; Shi et al., 2006; Viard, 2007; Park, 2010), in the software market (Larkin, 2008), and recently in insurance markets (Ericson, 2014; Miller, 2014). My paper adds to the recent extension of this literature to the health insurance sector, an important market featuring consumer switching cost.

The rest of this paper is organized as follows. Section 0.2 describes the empirical setting and data. Section 0.3 discusses important intuitions from relevant theory papers. Section 0.4 presents descriptive evidence and discusses alternative explanations. Section 0.5 lays out the structural model. Section 0.6 describes the empirical strategy and presents estimation results. Section 0.7 conducts counter-factual analysis of policy experiments. Section 0.8 concludes.

0.2 Empirical Setting and Data

0.2.1 Institutional Features of Medicare Part D

Medicare is a public health insurance program for the elderly and the disabled in the US. Medicare Parts A and B have covered hospital and physician services since the program's inception in 1965, but prescription drug coverage was not provided until the introduction of Medicare Part D in 2006. Providing outpatient prescription drug insurance to the elderly and the disabled, Medicare Part D is a large program in terms of both enrollment and spending. The Congressional Budget Office reports that in 2014, there were 37 million Medicare beneficiaries enrolled in Part D (Hoadley et al., 2014), and the Congressional Budget Office (CBO) estimates the program cost around 65 billion dollars⁴.

Unlike Medicare Parts A and B and other traditional government insurance programs, Part D is not delivered directly by the government, but rather by private insurers under contract with the government. These companies offer Medicare beneficiaries a choice between two types of prescription drug plans: bundled medical insurance and prescription drug benefits through the Medicare Advantage Prescription Drug plans (MA-PDs) that were in place prior to the deployment of Part D under Medicare Advantage⁵ and the stand-alone prescription drug coverage-only plans introduced in 2006. These stand-alone prescription drug plans (PDPs) are the focus of the present study.

Of all Medicare beneficiaries who have private prescription drug coverage, about 62 percent were enrolled in stand-alone prescription drug plans in 2012 (Hoadley et al., 2012). Stand-alone plans are offered in 34 geographically-defined markets within the continental United States. Plans in each market are offered by private insurers that are regulated by the government through the Centers for Medicare and Medicaid Services (CMS). In a typical market, approximately 20 firms offer more than 30 plans that are differentiated in terms of coverage. There are two broad types of prescription drug plans: basic plans that provide coverage actuarially equivalent to the required

⁴See Congressional Budget Office's Medicare Baseline Projection Reports in March 2015: http://www.cbo.gov/publication/44205.

⁵Medicare Advantage (MA) is a health insurance program of managed health care (preferred provider organization (PPO) or health maintenance organization (HMO)) that serves as a substitute for Medicare Parts A and B Medicare benefits.

minimum coverage as per the defined standard benefit set by the CMS and so-called "enhancedbenefit" plans that offer offer supplemental coverage on top of the minimum required coverage. Supplemental coverage relative to the defined standard benefit includes reduced deductible, partial or full coverage in the donut hole, reduced cost sharing, etc.

There are two types of Medicare beneficiaries, and I conceptualize the demand systems for both types based on the institutional setting in the structural model. Standard beneficiaries become eligible for Medicare at age 65. Enrollment takes place annually during an open enrollment period. After standard enrollees become eligible and first join Part D, they have to actively choose their prescription drug plans. In years after this initial enrollment, standard beneficiaries are defaulted into their previous plans unless they actively switch. Low-income enrollees are eligible through the low-income subsidy (LIS) system. Unlike standard beneficiaries, low-income enrollees do not need to choose their own plans or pay their own premiums and out-of-pocket costs. Instead, the government pays all or part of their premiums and out-of-pocket costs and randomly assigns them to basic plans priced below market average. Within both groups of beneficiaries, a small fraction of beneficiaries leave and a slightly higher fraction of new beneficiaries arrive annually: the annual attrition rate is around eight percent for standard enrollees and around ten percent for low-income enrollees; the annual arrival rate is around ten percent for standard enrollees and around thirteen percent for low-income enrollees.

Insurers can enter any market and offer one or more plans in each market. Within each market, price discrimination is not allowed and the same plan must be offered at the same price to both incumbent enrollees and newcomers. Premiums are set annually in two components – a basic premium for basic coverage, which applies to all plans, and a supplemental premium for supplemental coverage, which applies only to enhanced-benefit plans. Basic and supplemental premiums are set simultaneously, but in different manners. Supplemental premiums are set directly by insurers, while basic premiums are set through a centralized bidding process. Each year before the new enrollment cycle starts, insurers submit bids to the CMS for basic premiums. The CMS then computes the basic premiums for each plan as the insurer's bid minus the national average bid plus some base premium adjustments.⁶ This is referred to as a bidding process because basic plans that bid below market average win a share of low-income enrollees.

Insurer revenue is generated by enrollee premiums and three types of government subsidies. The government provides these subsidies to mitigate adverse selection and to partially insure insurers against excessive losses. First, plans are paid risk-adjusted subsidies based on each enrollee's health status or risk in terms of drug spending. Second, individual reinsurance covers 80 percent of catastrophic spending. Finally, risk corridors provide risk sharing between the government and insurers – excessive losses are partially compensated and excessive profits are taxed. Despite the complexity of the subsidy regime, variable profit or markup is still an increasing function of premiums, given any enrollee. In other words, the standard trade-off between a higher markup versus a higher market share still holds in this setting.

⁶Base premium is about one third of the enrollment-weighted national average bid.

0.2.2 Data

I use administrative data provided by the Centers for Medicare and Medicaid Services (CMS) on Medicare beneficiaries (henceforth "beneficiary files") and insurance plans (henceforth "plan files"). The beneficiary files cover a 20 percent random sample of Medicare beneficiaries from 2006 to 2011. For each year, this sample includes on average about 2.2 million standard enrollees and about 2 million low-income enrollees who are enrolled in stand-alone prescription drug plans. These beneficiary files include variables on enrollee demographics such as age, gender and race; on prescription drug plan choices in each year; and details on drug expenditures. The plans files include information on plan premiums and financial characteristics, such as a plan's deductible, gap coverage and tiered cost sharing. The recently released plan bridge files provide a crosswalk to unencrypted insurer and plan names, which allows me to identify the same insurer and plan across markets.

In descriptive evidence, I focus on a measure of markup or variable profit margin among standard enrollees to get rid of potential confounding variations in cost and subsidy that affect insurers' pricing decisions. I construct markups for each plan, averaged across its standard (low-income) enrollees, as the plan's premium plus the average risk-adjusted subsidy minus the expected cost, where the expected cost is defined as expected claims cost adjusted for pharmaceutical rebates, variable administrative cost and individual reinsurance. Details on the construction of expected claims cost will be discussed in Section 0.5.2. This markup measure incorporates the above-mentioned individual reinsurance and risk-adjusted subsidy. Individual reinsurance for catastrophic drug expenditure is computed using information on drug claims. To compute risk-adjusted subsidies in each year, I use the corresponding risk adjustment software from the CMS to compute the "risk score" for each enrollee. The CMS computes this risk score as a comprehensive summary of enrollee risk in terms of predicted drug spending and uses it to determine the amount of direct subsidy to pay insurers for each enrollee.⁷

Table 1 reports market-year level summary statistics. Panel A of Table 1 reports summary statistics on market structure, including numbers of insurers and plans, the Herfindahl index, and enrollment-weighted average premiums. The average number of number of insurers offering standalone plans in a market is 21, and there is meaningful variation across markets, ranging from 11 to 29. Most markets have more than 30 stand-alone plans, about half of which are basic plans and the other half of which are enhanced-benefit plans. There is also substantial cross-market variation in these numbers of plans. Part D markets are on average moderately concentrated, with a Herfindahl

⁷The CMS computes risk scores in each year using the corresponding software to predict each beneficiary's prescription drug spending in year t as a function of their inpatient and outpatient diagnoses from year t-1 and demographic information and uses these risk scores to determine risk-adjusted subsidy to insurers for each enrollee. To compute risk-adjusted subsidy in each year between 2006-2011, I use the corresponding RxHCC risk adjustment model from http://www.cms.gov/Medicare/Health-Plans/MedicareAdvtgSpecRateStats/Risk-Adjustors-Items/Risk2006-2011.html (retrieved June, 2014; last accessed October, 2015). Einav et al. (forthcoming) also use CMS software to compute risk scores as a proxy for individual predicted drug spending, and they use the 2012 model to compute risk scores for enrollees in 2006-2011 to consistently compare health status across years.

index of 0.22. There is some variation in premium levels across markets and a general increasing trend over time: enrollment-weighted average premium increased from 329 dollars in 2006 to 507 dollars in 2011.

Panels B and C of Table 1 report summary statistics on standard and low-income Medicare beneficiaries at the market level, including population size, annual attrition and arrival rates, and the share choosing stand-alone plans. Numbers of beneficiaries correspond to the 20% random sample, and should be multiplied by 5 to get actual Medicare population sizes. Arrival and attrition rates are relative to one-year lagged population sizes. Shares in stand-alone plans are calculated as out of the entire population of standard or low-income Medicare beneficiaries, including those with stand-alone plans, those with bundled coverage under Medicare Advantage, those with coverage provided by employers or third parties, and those without any prescription drug coverage. For both types of enrollees, the average attrition and arrival rates, as well as the share choosing standalone plans, do not vary much over time within each market. Therefore, I take these rates as constant for each market for the structural estimation.

Table 2 reports summary statistics on stand-alone plans, first pooled across years and then by year. The first row summarizes pooled data from 2006 to 2011, and each other row corresponds to a single year. The first column is the total number of plans. There were 1429 plans in 2006, and the decline in the number of plans over time is mostly driven by consolidations rather than exits. The second column reports numbers of plan entries, which were concentrated in 2006 and 2007. There were relatively few entrants overall after 2009. The third column reports numbers of plan exits, which are relatively low compared with the number of plans. Total premium is the annual total premium, which is the combined value of basic premium and supplemental premium.

As a simplification of my analysis, I focus on strategic pricing in this paper and abstract away from a second strategic response to inertia: since price discrimination across new and old enrollees is banned, firms face an incentive to continuously introduce new plans that can be priced low to "invest" in future demand while charging higher premiums to incumbent consumers. Many plans similar in coverage were forced to consolidate to comply with the "meaningful-difference" regulation, which was introduced in 2010 by the CMS to limit strategic entry behavior by requiring new plans to be sufficiently differentiated in coverage from existing plans by the same insurer. Although strategic plan entry is another important margins of firms' strategic behavior⁸, I abstract away from it and focus on pricing here – i.e., conditional on the set of plans being offered, withdrawn and consolidated each year, how do firms price their plans? I include all prescription drug plans, including both nonconsolidated and consolidated plans for the analysis. In order to link consolidated plans over time and to control for plan fixed effects for regression analysis later, I use each plan's most recent plan ID as its unique identifier.

⁸In a separate project, I document descriptive evidence consistent with such strategic entry and product proliferation.

0.3 Conceptual Framework

Klemperer (1987) discusses the general intuition for two counteracting incentives that firms face in the presence of consumer inertia, or when consumers behave as if they face switching costs. In the benchmark case without switching costs, demand in different periods is independent and so are firms' optimal strategies. However, when consumers face switching costs, demand is sticky over time, which creates two opposing incentives for firms: on one hand, firms want to charge low prices to "invest" in future demand, but on the other hand, firms want to charge higher prices to "harvest" inertial incumbent consumers. In a simple two-period model, firms only face the "invest" incentive in the first period and only face the "harvest" incentive in the second period. As a result, equilibrium price follows an "invest-then-harvest" pattern – firms charge low prices initially and increase prices afterward.

While a two-period model highlights the key trade-off firms face, a more realistic approximation of real markets is an infinite-period model, in which the invest and harvest incentives coexist. Beggs and Klemperer (1992) show how these two incentives interact in an infinite-period model with horizontal-differentiated products and consumers who are subject to switching costs. They solve for the unique Markov Perfect Nash Equilibrium under a critical assumption of perfect lock-in – i.e., consumers never switch because they are subject to infinitely large switching costs. In this equilibrium, prices are higher than the benchmark case without switching costs. This is not likely the case in a real-world context such as Medicare Part D, in which switching costs are not infinite as evidenced by the fact that some consumers do switch plans.

Dubé et al. (2009) relax this crucial assumption of infinite switching costs and examine an infinite-period model with switching costs and vertical differentiation. The authors establish the existence of a Markov Perfect Equilibrium and numerically solve for equilibrium prices as functions of switching costs. Applying this model to the markets of orange juice and margarine and using the empirically estimated level of switching costs in model simulations, they find equilibrium prices to be 18% lower than the case without switching costs, which reflects that the invest incentive dominates because of "the strategic effects of firms lowering their prices to defend themselves against other firms' attempts to steal customers". Moreover, the authors show that depending on the magnitude of switching costs, equilibrium prices can be higher or lower than the case without switching costs. For example, when switching costs are sufficiently large or even infinite, the harvest incentive dominates and prices are higher than the case without switching costs. In other words, with finite switching costs, it is an empirical question whether the invest or harvest incentive dominates. My supply model in Section 0.5 similarly features finite switching costs, and I will investigate this question empirically in the setting of Medicare Part D.

Applying these intuitions to the setting of Medicare Part D, plans only face the invest incentive when they first enter and face both the invest and harvest incentives in subsequent periods. Therefore, prices or markups should be lower when plans initially enter than in subsequent periods, and they should also be lower compared to incumbent plans. Because the invest and harvest incentives coexist every year except for the first, whether inertia leads to higher or lower equilibrium prices is an empirical question.

0.4 Descriptive evidence

0.4.1 Switching Costs

To lay the framework for my analysis of strategic pricing, I first summarize the existing evidence of consumer inertia and present some corroborative evidence. Polyakova (2015) documents evidence of consumer inertia in Medicare Part D, and estimates the magnitude of switching costs to be two to four times as high as average annual premiums. As corroborative evidence of this type of consumer inertia, Table 3 shows, separately for different cohorts of standard beneficiaries, the enrollment shares as of 2011 by plans introduced in each year. Note that most plans were introduced in 2006 and fewer plans were introduced in subsequent years, partly contributing to higher shares in plans introduced in 2006 all cohorts of enrollees. Interestingly, the percentage enrolled in the oldest plans (introduced in 2006) declines for younger cohorts of beneficiaries (84% among the 2006 cohort of enrollees and 72% among the 2011 cohort). Moreover, new cohorts of consumers are more likely to choose newly introduced plans. For example, the 2007 cohort is more likely than the 2006 and 2007 cohorts to choose plans introduced in 2006, and so forth. These statistics provide corroborative evidence that consumer inertia matters from the insurers' perspective.

0.4.2 Invest-then-Harvest Pricing

Figure 0-1, which displays enrollment-weighted markups over time for plans introduced in different years, shows two notable patterns. First, there is a general increasing trend: markups tend to increase as plans age. As I will show below, this increasing trend is robust across a variety of specifications. Second, within most years, entrants are priced lower than incumbent plans by a substantial margin. These patterns are consistent with the invest-then-harvest predictions discussed in Section 0.3.

To formalize this invest-then-harvest pricing pattern, I estimate regressions that compare markups between plans that have just entered the market and incumbent plans.

$$Markup_{kjmt} = \alpha + \beta \mathbf{1} \{Entry\}_{kjmt} + Coverage_{kjmt}\gamma + \delta_m + \lambda_t + \xi_k + \epsilon_{kjmt}$$
(1)

 $Markup_{kjmt}$ is markup averaged across standard enrollees for plan j offered by firm k in market m in year t. $1{Entry}_{kjmt}$ is a dummy indicating whether plan j first entered market m in year t. $Coverage_{kjmt}$ includes plan features, such as deductible level, whether the plan offers gap coverage and tiered cost sharing.

Table 4 reports ordinary least squares estimates of β from four specifications. Column (1) shows the raw correlation between markup and the entry dummy. Column (2) adds market and year fixed effects to account for potential differences across markets and over time that affect both entry and plan pricing. Column (3) adds insurer fixed effects to account for unobserved time-invariant heterogeneity at the insurer level. Column (4), which adds controls for plan coverage, compares plans within insurer and controlling for coverage. Consistent with the prediction in Section 0.3, these regression estimates suggest that the same insurer charges significantly lower markups on new plans than on incumbent plans with the same characteristics. Within the same insurer and year controlling for plan characteristics, markup is 148 dollars lower on entrant plans than incumbent plans, which is high relative to the average annual premium of 372 dollars.

0.4.3 Addressing Alternative Explanations

While the empirical invest-then-harvest pricing patterns documented in Section 0.4.2 are consistent with firms exploiting consumer inertia to maximize profits dynamically, such pricing patterns might also be rationalized by alternative explanations. First, since Medicare Part D is a new market, insurers might not be well-informed about cost, which could cause them to under-price initially and adjust prices upward as they learn about cost over time. Relatedly, in a learning-by-doing story, insurers might set low prices and invest in market shares in order to learn more quickly about cost. Finally, as Decarolis (2015) shows, the low-income-subsidy system also contributes to premium increases over time.

I have no intention of running a horse race to rule out these potential alternative explanations. It seems likely that insurer responses to inertia as well as these alternative stories are all empirically relevant to some extent. However, I argue that these alternative stories seem unlikely to explain the pricing patterns documented in Section 0.4.2. First, in the story of learning about cost, it is not clear why firms would systematically underestimate cost. Moreover, I find that even within the same firm, significantly lower markups are charged for entrant plans than for incumbent plans, contradicting the notion of learning about cost. Such within-firm comparisons also help contradict the learning-by-doing story.

In order to formally assess the robustness of my results to these alternative explanations, I test for evidence of the invest-then-harvest pricing pattern on subsamples of plans for which these alternative explanations are arguably not relevant. Table 5 summarizes these estimates for Equation 1 using subsamples. To address the first alternative explanation that firms learn about cost, the first three columns of Table 5 focus on subsamples of plans offered by insurers who are arguably well-informed about the cost of supplying prescription drug insurance to Medicare beneficiaries. Column (1) restricts the sample to plans offered by insurers that were major sponsors⁹ of Medicare Advantage prior to 2006 and that provided prescription drug insurance bundled with medical insurance to Medicare beneficiaries. Column (2) restricts the sample to plans offered by insurers with prior experience in Providing insurance coverage to Medicare beneficiaries. These three subsamples are not subject to the concern that insurers are not informed about cost. To address the second alternative explanation or the learning-by-doing story, Column (4) reports estimates

⁹Seven biggest sponsors in terms of market shares as of 2005 according to Gold (2006)

using a subsample of plans offered by insurers that are already experienced in Part D. Specifically, I assume that the benefit from such learning diminishes after the insurer serves many enrollees, which motivates restricting the sample of plans to those offered by insurers that have served at least 5000 enrollees before in the same market. Finally, to address potential confounding effects from the low-income-subsidy system, I use the subsample of enhanced-benefit plans, which are not eligible to receive low-income enrollees.

As shown in Table 5, the estimated coefficient-or the difference in annual markup between entrants and incumbent plans, holding the insurer and plan coverage as fixed, ranges from -\$134 to -\$187, which is not much different from the estimate of -\$148 on the full sample of plans. In other words, the result that markups are much lower on entrant plans than incumbent plans is robust to focusing on subsamples of plans where these alternative explanations are less relevant. This suggests that the empirical pricing pattern we observe is largely driven by strategic responses to consumer inertia rather than by these alternative explanations.

0.5 Model

My descriptive evidence in Section 0.4 rejects the null of no strategic response to inertia and is consistent with firms exploiting consumer inertia to maximize profits dynamically. To explore insurers' trade-offs between the invest and harvest incentives, I develop a dynamic model of insurers' pricing decisions that incorporates demand inertia and adverse selection. Structural estimation of this model in Section 0.6 uncovers insurers' discount factor, which quantifies the strength of the invest incentive. In Section 0.7 I further decompose observed pricing patterns to quantify insurers' trade-offs between invest and harvest incentives by comparing this model with a counter-factual benchmark with no inertia and with a counter-factual case where insurers are myopic and face no invest incentive. Finally, in Section 0.7 I simulate the price and welfare consequences of several policy experiments that could be used to constrain insurers' ability to exploit inertia.

0.5.1 Demand

As described in Section 0.2, there are two types of Medicare beneficiaries – standard enrollees and low-income enrollees – and I conceptualize the demand system for each type based on the institutional setting. I start with demand for standard beneficiaries, who make their own plan choices and are defaulted into their previous choices unless they actively switch. Since the focus of this study is on understanding firm pricing, I use the demand model and estimates from Polyakova (2015) for standard enrollees. In her model, standard enrollees are myopic¹⁰ and choose a plan

¹⁰One concern is that consumers can be forward-looking about changes in plan prices and their own health risk in the future. The latter is allowed by controlling for age, while consumers forward-looking about future price changes will be less likely to start with a cheap plan, which should dampen insurers' incentives to invest in market shares. However, as Handel (2014) argues, consumers make very poor decisions if we consider forward-looking demand. Moreover, dynamic demand adds additional complexity while dynamic supply is already computationally demanding. In fact, in dynamic games literature on durable goods, experience goods and network goods, it is fairly standard to

to maximize current utility, subject to switching costs. Let i denote individual, j plan, k insurer, m market (region) and t year. Individual i's utility¹¹ from choosing plan j in year t is as follows, where p_{kjt} is annual premium, ϕ_{kjt} is characteristics of the plan, and $1{\text{Default}}_{ikjt}$ is an indicator of whether consumer i is defaulted into plan j at time t. This default dummy is omitted for new enrollees, who are not defaulted into any plans.

$$u_{ikjt} = -\alpha p_{kjt} + \beta_{it} \phi_{kjt} + \gamma_{it} \mathbf{1} \{ \text{Default} \}_{ikjt} + \lambda_{it} \mathbf{1} \{ \text{Insurer} \}_k + \epsilon_{ikjt}$$
(2)

In this logit model, ϵ_{ikjt} is independent and identically distributed with a Type 1 Extreme Value distribution function.¹² ϕ_{kjt} includes the following characteristics that are feasibly observed by beneficiaries when they are making their choices: the deductible, the initial coverage limit, whether the plan offers coverage in the gap, whether the plan uses fixed dollar co-payments or coinsurance percentages, and whether the plan is eligible for getting low-income subsidy enrollees. Preferences over plan coverage β_{it} depend on the individual's demographics and health risk, $D_{it} =$ {age_{it},gender_i,race_i, risk score_{it}, esrd indicator_{it}}, where risk score is a measure of each beneficiary's health risk in terms of drug spending and esrd indicator is a dummy for end-stage renal disease. There are random coefficients in preferences over deductible, initial coverage limit and gap coverage: $\beta_{it} = \pi^{\beta} D_{it} + \psi_i^{\beta}$, where $\psi_i^{\beta} \sim N(\psi^{\beta}, \sigma^2)$. Switching costs γ_{it} and preference over insures λ_{it} also depend on individual demographics and health risk: $\gamma_{it} = \pi^{\gamma} D_{it} + \psi^{\gamma}$, and $\lambda_{it} = \pi^{\lambda} D_{it} + \psi^{\lambda}$.¹³

Standard consumer *i*'s probability of choosing plan *j* depends on plan features as well as the default plan *l*, which I denote as $P_j(p, l)$ and which follows the logit form. Active consumers without default plans face an unconditional choice probability $P_j(p)$. Aggregating individual choice probabilities, the share of standard enrollees choosing plan j in year t $S_{kjt}()$ is derived as follows.

$$S_{kjt}(p, \mathbf{S}_{t-1}) = \frac{1-\lambda}{1-\lambda+\mu} \sum_{l \in J(m)} S_{lt-1} P_j(p, l) + \frac{\mu}{1-\lambda+\mu} P_j(p)$$
(3)

 $S_{kjt}()$ is a function of prices p, lagged shares S_{t-1} , the attrition rate of standard enrollees λ and

assume myopic demand.

¹¹Polyakova (2015) points out "this formulation assumes that individuals choose the option with the highest "perceived" utility, which may not necessarily correspond to the highest "objective" valuation of plans as financial contracts (indeed, Abaluck and Gruber (2011, 2013) suggest that beneficiaries are choosing their plans inconsistently with the objective efficiency frontier)".

¹²Polyakova (2015) models choice among stand-alone prescription plans and does not include the outside option. I use a similar linear regression to predict, separately for incumbent and new beneficiaries, the share choosing to enroll in stand-alone plans instead of bundled coverage or no coverage, based on prices, number of plans, market fixed effects, etc. Estimates show that market fixed effect explains 66 percent for newcomers and 95 percent for incumbent enrollees. Details will be discussed in the Appendix. Alternatively, I can re-estimate the demand model using a nested logit model, in which beneficiaries first choose between not enrolling in prescription coverage, enrolling in bundled coverage and enrolling in stand-alone coverage, and then choose a plan if they choose any coverage in the first step. Since I focus on supply of stand-alone plans, I choose to take the simplistic approach instead to abstract away from the complexity of modeling demand for both bundled and stand-alone coverage.

¹³More specifically, preference over the two biggest insurers depend on D_{it} while preferences over other insurers follow the form of standard fixed effects.

the arrival rate μ . Because attrition and arrival rates do not vary much empirically within each market, I take λ and μ as exogenous and suppress their notations for this share function $S_{kjt}(p, \mathbf{S}_{t-1})$. Let J(m) denote the set of plans in market m. The first component of Equation 3 is the summation of shares across incumbent enrollees defaulted into different plans $(\sum_{l \in J(m)} S_{lt-1}P_j(p,l))$ weighted by the fraction of incumbent beneficiaries $(\frac{1-\lambda}{1-\lambda+\mu})$, while the second component represents the share among new beneficiaries without defaults (or the unconditional choice probability $P_j(p)$) weighted by the fraction of new beneficiaries $(\frac{\mu}{1-\lambda+\mu})$. In other words, other than prices, lagged market shares are important in determining current shares of standard enrollees because incumbent consumers' choice probability $P_j(p,l)$ is biased toward the lagged choice or default plan l. The importance of lagged shares is slightly depreciated by attrition of incumbent enrollees (λ) and arrival of new enrollees (μ) : empirically $\frac{1-\lambda}{1-\lambda+\mu}$ is approximately 0.90 in my data.

Although only around 20 percent of Medicare beneficiaries are low-income enrollees, they account for over 40 percent of enrollment in stand-alone prescription plans. Therefore, it is important to include profits from the population of low-income enrollees when modeling insurers' profit maximization problem. Unlike standard beneficiaries, low-income enrollees do not need to choose their own plans. Instead, the Centers for Medicare & Medicaid Services randomly assigns them to eligible plans when they first qualify for the low-income subsidy or when their previous plans are no longer eligible for receiving low-income enrollment. Low-income enrollees are evenly divided into eligible plans – basic plans priced below market average – except that an insurer eligible both in the last period and the current period keeps its incumbent low-income enrollees on top of this random assignment.¹⁴ Based on how low-income enrollees are automatically allocated across plans in reality, I model their discrete and mechanical demand, which depends on lagged low-income shares other than current plan bids for basic premiums.

Let λ^{LIS} denote the attrition rate of low-income enrollees and μ^{LIS} the arrival rate. The share of low-income enrollees assigned to basic plan j in year t $S_{kjt}^{LIS}()$ is computed as follows, where $\omega = \frac{1-\lambda^{LIS}}{1-\lambda^{LIS}+\mu^{LIS}}$ is the share of incumbent enrollees and $1-\omega = \frac{\mu^{LIS}}{1-\lambda^{LIS}+\mu^{LIS}}$ is the share of newcomers. The benchmark \bar{b}_{mt} is the average bid among basic plans weighted by lagged lowincome enrollment. $J_B(m)$ is the set of basic plans in market m. $N_{mt} = \sum_{l \in J_B(m)} \mathbf{1}\{b_{lt} \leq \bar{b}_{mt}\}$ is the number of basic plans pricing below benchmark. $S_{mt}^{Reassign}$ is the share of low-income enrollees

¹⁴In 2006, low income enrollees were randomly assigned to basic plans pricing below market average. In subsequent years, insurers keep previously assigned low-income enrollees conditional on having a basic plan pricing below the benchmark, where the benchmark is weighted by lagged low-income enrollment. Except for these enrollees who stay with a below-benchmark basic plan, low-income enrollees are again randomly assigned to basic plans pricing below benchmark. Low-income enrollees can choose to opt out of their assigned plans and choose a different plan and pay the difference in premiums. Among low-income enrollees choosing stand-alone prescription drug plans, the empirical fraction of "choosers" who have ever opted out increases over time from around 6% in 2006 to around 20% in 2010 (Summer et al., 2010). Such opting out behavior is not flagged in the administrative data, and cannot be identified except for those who choose plans not eligible for low-income enrollees. Decarolis et al. (2015) model the demand for such "choosers" based on the subsample for which opting out is observed in the data. I choose to model only the random assignment and not such opting out behavior because it is not essential to my focus of strategic pricing in response to inertia among standard enrollees.

who need re-assignment because their former insurers lost below-benchmark status.¹⁵

$$S_{kjt}^{LIS}(p, \mathbf{S}_{t-1}) = \begin{cases} 0 & b_{kjt} > \bar{b}_{mt} \\ (\omega S_{mt}^{Reassign} + 1 - \omega)/N_{mt} & b_{kjt} \le \bar{b}_{mt}, b_{kj't-1} > \bar{b}_{mt-1} \forall j' \in J_B(k) \\ \omega S_{kj,t-1}^{LIS} + (\omega S_{mt}^{Reassign} + 1 - \omega)/N_{mt} & b_{kjt} \le \bar{b}_{mt}, b_{kjt-1} \le \bar{b}_{mt-1} \\ \omega S_{kj',t-1}^{LIS} + (\omega S_{mt}^{Reassign} + 1 - \omega)/N_{mt} & b_{kjt} \le \bar{b}_{mt}, b_{kjt-1} > \bar{b}_{mt-1}, b_{kj't-1} \le \bar{b}_{mt-1} \end{cases}$$

$$(4)$$

In the first case in Equation 4, basic plan j prices above the benchmark and receives no lowincome enrollees. In the second case, basic plan j prices below benchmark, and the insurer k had no plans pricing below benchmark in the previous year. In this case, plan j receives an even share of incumbent low-income enrollees who need to be re-assigned ($\omega S_{mt}^{Reassign}$) plus new low-income enrollees $(1 - \omega)$. In the third case, basic plan j prices below benchmark, and it also priced below benchmark in the previous year. In this case, plan j receives an even share of randomly assigned enrollees as in the second case, while keeping its incumbent low-income enrollees ($\omega S_{kj,t-1}^{LIS}$). In the final case, basic plan j prices below benchmark, and another plan j' by the same insurer prices above benchmark but priced below benchmark in the previous year. In this case, plan j receives an even share of randomly assigned enrollees as in the second case, plus it keeps incumbent low-income enrollees within the same insurer ($\omega S_{kj',t-1}^{LIS}$).

0.5.2 Cost

Medicare Part D is a health insurance market with the potential for adverse selection. In my setting, health risk correlates with consumer preference as well as switching costs, as suggested by the demand estimates. Moreover, Handel (2014) and Polyakova (2015) both conclude that the interaction between adverse selection and switching costs has important welfare implications, which depend on the specific market setting. In order to account for this well-recognized issue, I follow Starc (2014) to model adverse selection and allow claims cost to be endogenous to price. This complication is a nuance rather than the focus of my model.

Conceptually, in the presence of adverse selection, consumers with different risks in terms of drug expenditure sort into different plans based on coverage and prices. As a result, insurers' claim costs depend on the types of consumers each plan gets, and therefore they are endogenous to price, which affects consumers' sorting behavior. I start with constructing the claims cost measure at the level of individual-plan pairs, before formulating endogenous claims cost at the plan level. For each enrollee's drug expenditure, an insurer is responsible for covering the remaining after subtracting the part paid out-of-pocket by the enrollee, the part covered by the government and the part rebated by the pharmacy and pharmaceutical manufacturers. As described in Section 0.2, the government provides three types of subsidies: risk-adjusted subsidies based on each enrollee's health risk, individual reinsurance for catastrophic drug spending, and risk corridors that partially

¹⁵This can be computed as $S_{mt}^{Reassign} = \sum_{l \in J_B(m)} \mathbf{1}\{b_{kj't} > \bar{b}_{mt}, \forall j' \in J_B(k)\}S_{klt-1}^{LIS}$.

compensate excessive losses and tax excessive profits. Individual reinsurance lowers insurer claims cost, while risk-adjusted subsidy and risk corridors do not enter claims cost directly but enter the profit function in other ways in Section 0.5.3. As for rebates, the claims data already incorporates rebates from pharmacies but not rebates form pharmaceutical manufacturers, which I will adjust for later using summary statistics from government reports.

Each individual's claims cost is constructed as total drug expenditure net of pharmacy rebates, individual reinsurance from the government and the enrollee's out-of-pocket spending. Intuitively, individual *i*'s claims cost to plan *j* offered by insurer *k* is a function of both plan coverage (X_{kjt}) and consumer demographics and health risk (W_{it}) . Insurer fixed effects δ_k are included to account for unobserved time-invariant heterogeneity in coverage or quality, such as broadness of pharmacy network, generosity of formularies and quality of customer service, which vary across insurers.

$$c_{ikjt}(X_{kjt}, W_{it}) = \alpha + X_{kjt}\beta + W_{it}\gamma + \delta_k + \xi_{ijt}$$
(5)

There are two important identifying assumptions embedded in this individual claims cost function. First, selection only works through observables. This is not a terrible assumption in this setting, as W_{it} includes enrollee risk score, which is a comprehensive risk measure in terms of expected drug spending. Second, this function assumes that there is no plan-individual specific moral hazard: while cost may depend on the plan's characteristics, the unexplained part of an individual's cost does not depend on the plan chosen. To the extent that the variation in plan coverage is well captured by both the detailed plan characteristics X_{kjt} and the insurer fixed effects included in the cost function, this assumption is justified because the notion of moral hazard in insurance markets typically refers to the fact that enrollees utilize more services with more generous coverage as they face a lower marginal price.

Based on this individual cost function and the demand system outlined in Section 0.5.1, I aggregate individual costs to get plan-level expected claims costs as follows. The cost (per enrollee) of plan j offered by insurer k depends on its coverage as well as the average characteristics of its enrollees, which is endogenous to price and the resulting selection.

$$C_{kjt} = \alpha + X_{kjt}\beta + E[W_{it}|\text{Choose Plan j}]\gamma + \delta_k$$
(6)

For tractability of the supply model, in which the state space includes lagged market shares by consumer type, I discretize types of standard enrollees based on risk score and gender.¹⁶ In other words, instead of controlling for enrollee characteristics W_{it} continuously in estimating the individual cost function 5, I drop W_{it} and estimate this function separately for each type of consumers. Within each type, consumers are assumed to be homogeneous (up to random coefficients in preferences) with cost realizations drawn from a common distribution. Low-income enrollees are taken

¹⁶Although the risk score is computed using demographics including gender, an OLS regression shows a small difference in cost by gender even conditional on risk score. To fully capture the cost difference across genders, I group standard consumers by gender in addition to risk score. I do not divide consumers by other demographic variables because they do not appear significant in predicting cost after controlling for risk score.

as homogeneous with a common cost distribution because of the automatic random allocation.

In order to get expected variable cost at the plan level, I adjust for two sources of variable cost other than expected claims cost. First, I take variable administrative cost to be 16% of claims cost based on estimates from other studies on similar markets: Starc (2014) estimates administrative cost to be 16% of premiums on average in Medigap; Ho et al. (2015) use data from the National Health Expenditure Survey to compute administrative cost to be 14-16% of total cost, and 16-19% of non-administrative cost, averaged across Medicare Advantage plans and plans in Medicare Part D. Second, I take rebates from pharmaceutical manufacturers to be 10% of total drug spending based on summary statistics from government reports: Boards of Trustees (2012) reports that the average manufacturer rebate rate, as a percentage of total prescription drug costs, ranged between 8.6 percent and 11.3 percent between 2006 and 2010.¹⁷

Other than expected variable cost, cost realizations also matter for insurers' dynamic profit maximization because of risk corridors. Risk corridors provide risk sharing between the government and insurers by partially compensating excessive losses and taxing excessive profits. In order to account for this when estimating the supply model, I randomly draw realized cost from normal distributions centered around expected cost and average across these random draws to get expected insurer profit. The standard deviation of this distribution of plan-level cost is calculated using standard deviation of individual cost and plan enrollment.

0.5.3 Supply

My model of insurers' strategic response to inertia in pricing builds on the work of Beggs and Klemperer (1992) and Dubé et al. (2009) and incorporates new features based on my empirical setting. As in Dubé et al. (2009), I consider an overlapping generations model with imperfect lockin. In this model, both single-product and multi-product insurance firms offer differentiated plans and compete for consumers subject to switching cost. In each period, a fraction of old consumers leave the market and new consumers arrive.

In order to focus on insurers' dynamic pricing decisions in the presence of inertia, I make the following simplifying assumptions. First, I take plan characteristics as given, which is not a bad approximation in my setting, as empirically insurers tend to adjust premiums instead of plan characteristics. Second, I take market structure as given and abstract away from strategic entry. This assumption is less innocuous because entry does happen empirically. Since price discrimination is not allowed, firms face an incentive to continuously introduce new plans that can be priced low to invest in future demand while also charging higher premiums to incumbent consumers. However, the "meaningful-difference" regulation essentially put an end to such strategic entry, and the number of plans remains quite stable afterward. Although the timing of this regulation is close to the end of the sample period, it suffices in confirming that there will not be unobserved entry after the sample period, as the supply model involves forward simulation for many more periods. In other words, entry is less common in recent years and will continue to be less common looking-ahead. Relatedly,

¹⁷There is no need to adjust for pharmacy rebates, which are already net out in the claims data.

I only model variable profits of insurers and not fixed costs, which are sunk costs and therefore not relevant for pricing decisions. Third, I assume that insurers take the regulation environment as given, without foresight of future policy changes. Finally, I assume pricing decisions are separately made for stand-alone prescription drug plans and Medicare Advantage plans that bundle medical insurance and prescription benefits. Although cannibalization between these two segments is a concern, it is not essential to the invest-then-harvest pricing story, and I focus on the pricing of stand-alone prescription drug plans and abstract away from modeling the demand and supply for MA-PD plans.

0.5.3.1 Value Function

Insurers account for consumer inertia and choose bids and supplemental premiums to maximize discounted profits. As calculated in Equation 7, $V(\sigma_k, \sigma_{-k}, \delta, \mathbf{SV}_{mt_0})$ is the expected present value of profit for firm k in market m in year t_0 , where σ denotes pricing strategies, δ denotes insurers' annual discount factor, SV includes the state variables and Π denotes annual variable profit.

$$V(\sigma_k, \sigma_{-k}, \delta, \mathbf{SV}_{mt_0}) = \mathbb{E}\left[\sum_{t=t_0}^{\infty} \delta^t \Pi_{kt}(\sigma_k, \sigma_{-k}, \mathbf{SV}_{mt})\right]$$
(7)

Since firms account for demand inertia, profits and pricing strategies are state-dependent. Besides exogenous state variables, including plan characteristics and enrollee characteristics, because of inertia (Beggs and Klemperer, 1992), SV_{mt} also includes lagged market shares by consumer type, which evolve deterministically based on the demand system in Section 0.5.1. Insurer profits and pricing strategy depend on lagged market shares among both standard and low-income enrollees. First, lagged standard enrollee shares matter because, intuitively, the harvest incentive depends on how many standard enrollees an insurer has locked in. Second, lagged low-income enrollee shares also matter for insurer profits and pricing strategy due to the way low-income enrollees are assigned as described in Section 0.5.1. In the presence of adverse selection, different types of enrollees differ in cost and demand. Therefore, lagged shares of different types of enrollees arguably affect insurer pricing differently, and I include lagged shares by consumer type to account for this. As a side note, lagged market shares by consumer type pin down expected cost, and therefore there is no explicit cost term in the value function.

Insurer k's pricing strategy σ_k for all its plans $(j \in J(k))$ is a mapping from states \mathbf{SV}_{mt} to bids for basic premiums (b_{jt}) and supplemental premiums (PS_{jt}) . More specifically, σ_k includes bids for basic premiums for each plan $b_j(\mathbf{SV}_{mt}, \epsilon_{jt}) = f(\mathbf{SV}_{mt}) + \epsilon_{jt}$ and supplemental premiums for each enhanced-benefit plan $PS_j(\mathbf{SV}_{mt}, \epsilon_{jt}) = h(\mathbf{SV}_{mt}) + \epsilon_{jt}$.¹⁸

¹⁸The interpretation for ϵ_{jt} is managerial mistake or specification error, and is assumed to be drawn independently across plans and years from a normal distribution centered around zero.

0.5.3.2 Annual Profit Function

Firm k's annual variable profit consists of profits from all its plans in market $m, j \in J_m(k)$. In other words, multi-product firms jointly maximize profit across all plans. Plan j's total profits include profits from different groups of enrollees Π_{kjt}^{θ} , where θ represents discrete types of standard enrollees and the group of low-income enrollees. $\Gamma(.)$ is a function representing adjustments from the risk corridors, which partially compensate for excessive losses and tax excessive gains.

$$\Pi_{kt}(b, PS, \mathbf{SV}_{mt}) = \sum_{j \in J(k)} \Gamma(\sum_{\theta} \Pi^{\theta}_{kjt}(b, PS, \mathbf{SV}_{mt}))$$
(8)

Plan j's (pre-risk-corridor) profit from each enrollee type can be calculated as enrollment times markup.

$$\Pi^{\theta}_{kjt}(b, PS, \mathbf{SV}_{mt}) = M^{\theta}_{mt} S^{\theta}_{kjt}(b, PS, \mathbf{S}_{t-1}) Markup^{\theta}_{kjt}(b, PS, \mathbf{SV}_{mt})$$
(9)

 M_{mt}^{θ} denotes the population of each type of enrollee within market m in year t. S_{it}^{θ} denotes shares of each type of enrollee choosing plan j in year t, which is calculated based on the demand system as shown in Equations 3 and 4. Markup on each enrollee type is equal to total premium minus expected cost plus risk-adjusted subsidy. Expected cost is constructed in Section 0.5.2, while premiums and subsidies are computed following the actual process of setting prices and government subsidy. Each year before enrollment takes place, for each plan $j \in J_m(k)$, insurer k submits a bid b_{jt} for its basic premium and sets directly the supplemental premium PS_{jt} if it is a enhanced-benefit plan. The CMS computes basic premium as $PB_{jt} = b_{jt}$ (national average bid – base premium), where base premium is a fixed fraction of national average bid. Enrollees face a total premium $p_{jt} = PB_{jt} + PS_{jt}$, where $PS_{jt} = 0$ for basic plans. In order to mitigate adverse selection in this market, the government computes a risk score for each enrollee r_{it} , based on demographics and medical history, and pays risk-adjusted subsidy $r_{it}b_{jt} - PB_{jt}$ to the insurer. For an average enrollee with a risk score of one, the sum of basic premium and risk-adjusted subsidy is equal to the bid for basic premium. In other words, although risk-adjusted subsidy is endogenous to plan bids for basic premiums, insurers still face the standard trade-off between a higher markup (as a result of both a higher enrollee premium and a higher government subsidy) and a higher market share when setting prices.

I restrict insurers' strategies to be Markovian because the full set of dynamic Nash equilibria is unbounded and complicated. The Markov-Perfect Nash Equilibrium requires $V(\sigma_k, \sigma_{-k}, \delta, \mathbf{SV}_{mt}) \geq$ $V(\sigma'_k, \sigma_{-k}, \delta, \mathbf{SV}_{mt})$ given competitors' strategies σ_{-k} for all states and alternative strategies σ'_k , i.e. each insurer's strategy has to be optimal given competitor's strategies.

0.6 Structural Estimation

0.6.1 Demand Estimation

Table 6 reports Polyakova (2015)'s simulated maximum-likelihood estimates on a few important demand parameters. Estimates for the price coefficient and the switching cost dummy are relatively robust across specifications. Besides including a rich set of plan features in all specifications, Columns (3) and (4) include more insurer fixed effects than the first two columns¹⁹ to address the concern with unobserved insurer quality affecting both premiums and demand. Moreover, Columns (2) and (4) use lagged cost as an instrument for plan premium to address the concern with unobserved plan quality affecting both premiums and demand. Both instrumenting and controlling for more insurer fixed effects only increase the magnitude of the premium coefficient slightly, which confirms that including rich plan characteristics leaves little room for unobserved insurer and plan quality to affect both pricing and demand.

Controlling for more insurer fixed effects reduces the magnitude of the intercept for the switching cost term from 5.6 to 5.1, or reduces the implied switching cost for a 75 year-old female enrollee with average risk from \$1330 to \$1164. This difference suggests that there is unobserved quality at the insurer level that enrollees persistently value over time, and it is important to account for those unobservables with insurer dummies. Therefore, I choose the last specification with instruments for premium and ten insurer fixed effects as input for my supply estimation.

0.6.2 Cost Estimation

Figure 0-2(a) visually summarizes the individual cost estimation results. This figure reports, for each type of enrollee, expected cost to a basic plan offering minimum coverage and to an enhancedbenefit plan with more generous coverage (zero deductible and gap coverage). Standard enrollees are divided into groups with low, medium and high risk scores. In addition to these expected cost measures, Figure 0-2(b) adds switching cost in dollars and willingness-to-pay for more generous coverage. There is a significant correlation between cost and willingness-to-pay for extra coverage across different types of plans. There is also a small positive correlation between cost and switching cost, but this is less strong than the correlation between cost and willingness-to-pay.

Figure 0-2 pools female and male enrollees for simplicity, and Appendix Figure A.1 also breaks down by gender in addition to risk score. The patterns look similar – there is a lot of cost heterogeneity across enrollee types, and cost correlates strongly with preference and weakly with switching cost. More details on the estimation results are reported in Appendix Table A.1.

¹⁹Columns (1) and (2) include three insurer fixed effects by including dummies for the two biggest insurers (the omitted category consists of all other insurers), while Columns (3) and (4) include three insurer fixed effects by including dummies for the nine biggest insurers (the omitted category consists of all other insurers).

0.6.3 Supply Estimation

Estimating parameters of dynamic games and computing equilibria are computationally demanding (Benkard, 2004; Bajari et al., 2010). The large number of insurers in Part D markets makes it even more difficult computationally. Instead of solving for the equilibrium of the supply model, I follow Bajari et al. (2007)'s two-step approach to uncover insurers' valuation of future profits. Essentially, this approach minimizes the violation of insurer rationality by finding the parameter value or insurers' discount factor such that the observed pricing strategies are closest to equilibrium strategies. This approach is implemented in two steps. In the first step, I empirically estimate how insurers price their plans by regressing premiums on relevant state variables. Such reduced-form estimates empirically correlate insurers' actions to states and characterize insurers' strategies $\sigma(SV)$, which are also referred to as the empirical policy functions.

In the second step, I take competitors' strategies as given by these empirical strategies characterized in the first step and forward simulate to construct insurers' discounted profits $V(\sigma_k, \hat{\sigma}_{-k}, \delta, \mathbf{SV})$ as in Equation 7 given a discount factor. This simulated value function can be constructed using both each insurer's empirical strategy and alternative strategies. Imposing rationality or optimality on insurers' decisions based on the definition of MPNE in Section 0.5.3, I estimate the discount factor δ such that profitable deviations from empirical policies are minimized, i.e. the empirical strategies reflect minimum violation of rationality.

In other words, I assume the insurers solve the dynamic pricing game in Section 0.5.3 and set their pricing strategies accordingly, and I look for parameters of the supply model such that insurers' pricing behavior is optimal. Besides model assumptions in Section 0.5.3 and the following functional form assumption in Section 0.6.3.1, this estimation approach requires that insurers in all markets play the same equilibrium strategies so that data from all markets can be used to jointly characterize empirical pricing strategies in the first step.

0.6.3.1 Step One: Empirical Pricing Policy Function

I let the data reveal insurers' empirical pricing strategies by estimating prices or premiums as functions of shares as well as other determinants of pricing decisions as in equation 10. Premiums p_{jt} include bids for basic premium b_{jt} for basic plans and supplemental premiums PS_{jt} for enhancedbenefit plans. The controls include own lagged shares by enrollee type S_{kjt}^{θ} and shares of other plans offered by the same insurer S_{k-jt} , plan characteristics X_{jt} , and insurer fixed effects to account for unobserved heterogeneity across insurers that affect both shares and pricing decisions. The residual is assumed to be normally distributed, and I use the estimated standard deviation to get random draws for competitors' prices for forward simulations in the second step.

$$p_{kjt} = \alpha + \sum_{\theta} \beta^{\theta} S^{\theta}_{kjt} + \sum_{\theta} \gamma^{\theta} S^{\theta}_{k-jt} + X_{jt} \lambda + \xi_k + \epsilon_{jt}$$
(10)

These empirical policy functions condition on a coarser set of state variables than what is required to compute a Markovian strategy and are similar in nature to the notion of oblivious strategy as formalized by Weintraub et al. (2008). As an approximation for Markov perfect equilibria, Weintraub et al. (2008) define oblivious equilibrium as an equilibrium in which each firm is assumed to make decisions based on its own state and knowledge of the long-run average industry state. The rationale for using a coarser set of state variables in my setting is the same as that for computing oblivious equilibrium: realistically it is computationally infeasible to compute Markov perfect equilibria when market sizes are large and the state space explodes even with 20 firms. Such simplifications can actually provide good approximation to firms' equilibrium behavior. In fact, Weintraub et al. (2008) show that the oblivious equilibrium approximates a Markov perfect equilibrium as the number of firms grows.

I estimate empirical pricing strategies separately for three clusters of plans: basic plans offered by single-product firms, basic plans offered by multiple-product firms, and enhanced-benefit plans offered by multiple-product firms. Different factors are relevant for pricing across these clusters of plans – for example, controls are different for single- versus multiple-product firms (shares of other plans within firm are not relevant for the former). Therefore, I estimate the empirical pricing functions separately for these three clusters of plans.

Table 7 summarizes my key coefficient estimates. Not surprisingly, plans with higher coverage are more expensive: premiums decrease with deductible amount and increase with gap coverage. Premiums also depend on lagged shares, but the coefficient varies across clusters of plans and types of enrollees. Finally, the key takeaway is that the adjusted R^2 is reasonably high, meaning that this first step is doing a good job at predicting what firms do based on these observable factors, which is a prerequisite for feeding these empirical policy functions into the second step to estimate firms' discount factor.

0.6.3.2 Step Two: Uncover Insurers' Discount Factor

Given the discount factor and pricing strategies, I can forward simulate to get the empirical value function for insurer k, the empirical counterpart to the value function in Equation 7.

$$\hat{V}(\sigma_k, \sigma_{-k}, \delta, SV_{mt_0}) = \mathcal{E}_n[\sum_{t=t_0}^{\infty} \delta^t \Pi_{kt}(\sigma_k, \sigma_{-k}, \mathbf{SV}_{mt}) | \mathbf{SV}_{mt}]$$
(11)

I take competitors' strategies σ_{-k} as given by empirical pricing strategies estimated from the first step and consider each insurer's optimization problem separately. In order to compute this simulated value function for each possible σ_k , including the empirical strategy and alternative strategies, I forward simulate 500 times and take the average across simulations to get \hat{V} . The discount factor can be estimated using the simulated minimum distance estimator as follows, where N is the number of states times the number of alternative-strategies considered.

$$\hat{\delta} = \operatorname{argmin} \frac{1}{N} \sum_{\tilde{\sigma}_k, \mathbf{SV}_{mt}} (\min\{\hat{V}(\tilde{\sigma}_k, \hat{\sigma}_{-k}, \delta, \mathbf{SV}_{mt}) - \hat{V}(\hat{\sigma}_k, \hat{\sigma}_{-k}, \delta, \mathbf{SV}_{mt}), 0\})$$

Intuitively, the discount factor reflects minimum violation of insurer rationality by minimizing

room for profitable deviations. The objective function is the average forgone profit by choosing empirical strategies $\hat{\sigma}_k$, compared with alternative strategies $\tilde{\sigma}_k$. Since MPNE requirement applies to all possible alternative strategies, alternative strategies can be any perturbations of empirical strategies. Therefore, I consider single-period deviations from the empirical policy functions for simplicity and consider 100 alternative strategies for each insurer.

Conceptually, the discount factor tells us how much insurers care about future profits and therefore how strong the invest incentive is. The identification comes from the observed price or markup levels – intuitively, the more insurers care about the future, the stronger invest incentive they face and the lower they set the premiums. Table 8 reports the estimated $\hat{\delta}$ of 0.946²⁰, which suggests that insurers value future profits strongly and therefore face a strong invest incentive. The standard error is bootstrapped.

0.7 Counter-factual Analysis

Section 0.4 shows a striking invest-then-harvest pricing pattern that is consistent with insurers exploiting consumer inertia. Structural estimation in Section 0.6 uncovers a high discount factor, indicating that insurers have very strong incentives to invest in future demand. Should we worry about such invest-then-harvest pricing among Part D sponsors? On one hand, price increases over time create dynamic choice inefficiency in consumer choice in the presence of inertia. On the other hand, the net effect on consumer welfare also depends on whether switching costs toughen or soften competition. In this section, I apply the dynamic supply model above to answer two important economic questions. First, what is the net effect of strategic pricing in response to inertia on equilibrium prices? This is an empirical question and depends on whether the invest or the harvest incentive dominates. To quantify insurers' trade-off between these counteracting incentives, I decompose the observed pricing patterns into components attributed to invest and harvest incentives in Section 0.7.1. Second, what are consequences of policies that could be used to constrain insurers' ability to exploit inertia using the "invest-then-harvest" pricing tactic? To evaluate the desirability of government intervention, I simulate the effects of three policies on prices and welfare, including two policies implemented or proposed under the Affordable Care Act.

0.7.1 Do Switching Costs Lead to Higher or Lower Prices?

The competitive effect of switching costs is ambiguous and depends on whether the invest incentive or the harvest incentive dominates. While Beggs and Klemperer (1992) show that the harvest

²⁰One potential concern is that such a high annual discount rate cannot be reconciled with the fact that many Part D sponsors are publicly traded and have high rates of returns on investment. However, it should be noted that I estimate a common discount rate for all insurers in this market for computational feasibility. While the discount factor or rate of return might vary across insurers empirically, this estimate represents the average discount factor across insurers. Furthermore, even for big insurers such as Humana, the annual rate of return is not much higher than that implied than the estimated discount factor. For example, Humana's recent annual return on investment ranges from 6.37% to 7.91% based on http://csimarket.com/stocks/HUM-Return-on-Investment-ROI.html.

incentives always dominates when consumers are perfectly inertia and switching costs are infinite, this is not necessarily true when consumers are subject to finite switching costs. In fact, Dubé et al. (2009) show that depending on the magnitude of switching costs, equilibrium prices can be higher or lower than the case without switching costs. Contrary to conventional wisdom that switching costs soften competition, the authors show examples where inertia reduces equilibrium prices. When switching costs are finite, firms face incentives to price low not only to attract new consumers but also to attract consumers currently attached to competitors. Arie and Grieco (2014) highlight the "compensating" effect, or the incentive to induce competitors' consumers to switch products, as the key contributing factor to lower price levels.

In order to decompose the effects of the invest and harvest incentives on driving prices, I compare prices in the model with inertia with two counter-factual benchmarks, one without inertia and one in which insurers are myopic. In the dynamic model with inertia, insurers are subject to both the invest and harvest incentives when setting prices. In the counter-factual benchmark with no inertia, insurers are subject to neither the invest incentive nor the harvest incentive. In the counter-factual with myopic insurers, insurers face no invest incentive and only the harvest incentive. The comparison between the dynamic model with consumer inertia and these two counter-factual benchmarks helps decompose insurers' trade-off between the invest and harvest incentives.

In the counter-factual benchmark with no inertia, standard enrollees' demand is different from Section 0.5.1 because their utility, which is described below by Equation 12, no longer includes switching costs as in Equation 2.

$$u_{ikjt} = -\alpha p_{kjt} + \beta_{it} \phi_{kjt} + \lambda_{it} \mathbf{1} \{ \text{Insurer} \}_k + \epsilon_{ikjt}$$
(12)

In the counter-factual with myopic insurers, demand is the same as in Section 0.5.1, but now the discount factor $\delta = 0$ in the supply model, and insurers set prices only to maximize annual variable profits without any consideration for future profits.

Table 9 reports enrollment-weighted equilibrium markups among standard enrollees in a simplified two-period model with inertia corresponding to the actual setting, in the counter-factual benchmark with no switching costs and in the counter-factual with myopic insurers. Consistent with the invest-then-harvest intuition, in the model with inertia we see low prices (small and negative average markup) in the first year but high prices (high average markup) in the second period. Interestingly, average markup is lower than the benchmark with no inertia, which indicates that the invest incentive dominates the harvest incentive and that inertia toughens competition. A comparison between the model with inertia and the counter-factual benchmark without inertia in the first year shows that the invest incentive accounts for a drop of around \$300 in markup. Another comparison between the model with inertia and the counter-factual with myopic insurers in the second year shows that the harvest incentive accounts for an increase of around \$100 in markup. These comparisons show that the invest incentive dominates the harvest incentive and that switching costs make the market more competitive. These findings contradict the conventional wisdom that switching costs soften competition, and confirm the conclusions of Dubé et al. (2009) and Arie and Grieco (2014).

0.7.2 Policy Experiments

The effects of government intervention are not directly intuitive and are in fact ambiguous ex-ante, because pricing response to inertia creates two offsetting effects on consumer welfare. On one hand, price increases create dynamic choice inefficiency in consumer choice in the presence of switching frictions. On the other hand, the structural estimation suggests that inertia reduces prices as insurers face very strong incentives to invest in future profits. The desirability of government intervention depends on how effectively each policy can smooth prices over time without increasing average price levels. In this section, I apply my model to understand the potential role of government intervention by simulating the price and welfare consequences of policy experiments where the government restricts insurers' ability to exploit consumer inertia with the invest-then-harvest pricing tactic.

First, the most straight-forward way to constrain insurers' invest-then-harvest pricing strategy is to cap the annual increase in plan bids and supplemental premiums by a certain percentage. This cap directly curbs insurers' ability to harvest and raise prices later, and therefore also dampens the invest incentive upon entry. In fact, the Affordable Care Act implements an "Effective Rate Review" policy closely resembling a cap on annual premium increase: "any proposed rate increase by individual or small group market insurers at or above 10 percent will be scrutinized by independent experts to make sure it is justified" (U.S. Department of Health and Human Services, 2014). Motivated by this policy, I consider a policy experiment in which firms can only increase bids and supplemental premiums by up to 10 percent each year. The model is set up similarly to that in Section 0.5, except that now there is a constraint that insurers' bids and supplemental premiums cannot exceed 110 percent of those in the previous year.

A second policy I consider is to offer a public option at a low price to compete with private insurers. Widely discussed in privatized insurance markets, public options were proposed as part of the Affordable Care Act but were removed in the final reconciled bill. Intuitively, offering an inexpensive public option increases competition, which restrains room for insurers to harvest consumer inertia or charge high prices later on and, as a result, also weakens the invest incentive early on. I consider a policy experiment in which the government adds a public option to the market, offering the minimum required coverage and priced at \$300 in all years. The model is similar to that in Section 0.5, except that now there is additional competition from this public option.

Finally, I also consider the effects of removing risk sharing and fully exposing insurers to excessive losses and gains from their pricing decisions. The risk corridors might have exacerbated insurers' invest-then-harvest pricing tactic by making it less costly for insurers to price low initially to attract consumers. However, risk corridors might also weaken the invest-then-harvest incentives because insurer profits exceeding a threshold are taxed. The net effect is ambiguous and is an empirical question. Given the importance of the risk corridors in this setting, I analyze this counter-factual to understand the effect of this regulation (or its removal) on insurer pricing and consumer welfare. The supply model remains the same as in Section 0.5, except that risk corridor adjustment Γ is removed from the profit function in Equation 8. A caveat with the last two policy experiments is that public options and risk corridors are big policy instruments with many potential effects other than influencing insurer response to inertia, and my analysis here only speaks to one of many aspects of their effects.

0.7.3 Implementation

The empirical policy functions estimated in Section 0.6.3.1 only characterize equilibrium strategies in the empirical setting and can no longer be used as competitors' strategies in the counter-factuals. Instead, I need to solve for the Markov Perfect Nash Equilibrium in each counter-factual. It is computationally difficult to solve for the equilibrium in each game, given the large number of insurers and the large number of parameters to solve for in the equilibrium strategy. For computational feasibility, I restrict the set of strategies to follow the functional form of the empirical policy function. I assume that in each counter-factual, equilibrium pricing strategies (bids for basic premiums and supplemental premiums) take the functions form of the following, where I constrain the coefficients on shares to change by the same proportion relative to the empirical coefficients and constrain the coefficients on plan characteristics to change by the same proportion relative to the empirical coefficients.

$$\tilde{p}_{kjt} = \tilde{\alpha} + \tilde{\xi}_k + \sum_{\theta} \tilde{\beta}^{\theta} S^{\theta}_{kjt} + \sum_{\theta} \tilde{\gamma}^{\theta} S^{\theta}_{k-jt} + X_{jt} \tilde{\lambda} + \epsilon_{jt}$$
(13)

With this simplification, I forward simulate to get the expected value functions given the price vector in the first year P_0 and the coefficients that guide pricing strategies in subsequent periods. I then iterate over insurers' optimal choices of initial prices and these parameters using the simulated value functions until a fixed point is reached, which provides the equilibrium pricing strategies. Given the solved equilibrium strategies, I move on to Section 0.7.4 and compute welfare.

For computational feasibility, I conduct counter-factual analysis on one representative market with around a quarter of a million enrollees annually choosing stand-alone plans.²¹ The population sizes of both standard and low-income enrollees are close to cross-market averages as reported in Table 1. In addition, for computational feasibility given the large number of insurers, I report results from simplified two-period models. Although the price levels would be more comparable to the data in the model with a longer time horizon, two-period models already highlight intuitions for the key economic forces, and the following qualitative interpretations are not an artifact of the two-period set up.²² As a benchmark for comparison, I also solve for the equilibrium in a two-period

²¹Although I only perform the simulations on one representative market, I assume the same policy experiment is implemented throughout all markets and the national-average bid (used to transform bids to premiums and subsidies) changes by the same proportion as in this market.

²²While the two-period models suffice in illustrating the key insights, I am working on models with a longer time horizon in order to verify the robustness of the conclusions and to provide a more realistic comparison with the data.

game with the current setting, i.e. with consumer inertia, no cap on price increase, no public option and with the current risk corridor set up.

0.7.4 Welfare Metrics

I evaluate effects of three policy experiments on prices as measured by enrollment-weighted premiums and markup levels, as well as on social welfare, including consumer welfare, insurer profit and government subsidy. I compute consumer welfare, insurer profit and government subsidy as relevant to standard enrollees because the automatic allocation of low-income enrollees makes it difficult to infer their preferences and to compute their welfare. In order to consistently compare static and dynamic counter-factuals, I define these welfare measures on a per-period basis. Total surplus W on a per-period basis can be calculated as follows, where CS denotes consumer surplus in money metrics, II denotes total insurer profits, Subsidy denotes government subsidy, and λ is the social cost of raising public funds. I take $\lambda = 1.3$, based on estimates in Hausman and Poterba (1987).

$$W = CS + \Pi - \lambda Subsidy \tag{14}$$

Whether consumer surplus should include switching costs depends on the underlying causes for observed inertia (Handel, 2014, provides a detailed discussion). Such distinction is not crucial for welfare analysis in my policy experiments, which all directly influence the supply side rather than the demand side. Here I take switching costs as welfare-neutral, but when evaluating the robustness of my results, I plan to consider switching costs as partly or fully welfare-relevant. When treated as welfare neutral, switching costs do not count towards actual utility but do affect choice probabilities. I simulate individual utilities and choices to compute consumer welfare in monetary terms after the normalization over the absolute value of the price coefficient.

Insurer profits are calculated based on individual choice simulations above and using the relevant profit functions. I compute pre-risk-corridor profits because risk corridor payments will be reported separately as part of government subsidy. There are three types of subsidies relevant to standard enrollees – risk-adjusted direct subsidy for each enrollee to insurers; risk corridor payments to insurers in cases of excessive losses (but the payment can go the other way if the insurer earns excessive profits); and individual reinsurance to insurers, which covers 80 percent of catastrophic drug expenditures. In the counter-factuals, I focus on direct subsidy and risk corridor payments, which are endogenous to firms' pricing strategies. These subsidies can be computed similarly to the profit term above on an per-year basis. The other subsidy, individual reinsurance, is not likely to change much across counter-factuals and therefore is less interesting for the counter-factual exercise. Following Decarolis et al. (2015), I compute government subsidy as relative to what would have been spent subsidizing the same individuals in MA-PD instead, assuming that in the absence of stand-alone plans, an enrollee would get prescription drug coverage through MA-PD plans instead.

0.7.5 Policy Experiment Results

Table 10 reports enrollment-weighted equilibrium markups among standard enrollees, first predicted by the model with inertia corresponding to the actual setting as the benchmark for comparison, then from three policy experiments: setting a cap on the percentage of annual premium increase, providing a public option at a low price, and removing risk sharing between the government and insurers. All three policies dampen the invest incentive in the first year and lead to higher markups on average, to different degrees. All three policies also dampen the harvest incentive in the second period and lead to lower markups on average, to different degrees.

In the counter-factual with a 10 percent cap on annual bid and supplemental premium increase, average markup is higher relative to the benchmark case in the first year, but is lower in the second year, which is consistent with the intuition that such a policy dampens the invest-then-harvest incentives. The average price level is slightly lower than the benchmark without any policy experiment. In the counter-factual where a public option is offered at a fixed price of \$300, markups in both periods are lower compared to the benchmark, but price in the second year is still much higher than that in the first year. In the last counter-factual, where the risk corridor is removed, prices are only slightly different from the benchmark, if noticeable. To sum up, capping annual premium rise is the most effective in terms of both smoothing prices over time and constraining average price levels. Offering a low-price public option constrains markup rise in the second year but not by much. Removing the risk corridors has little impact on markups.

The prediction that the cap on annual premium increase rate can lower average prices is not as intuitive as the other prediction on smoothing pricing dynamics. On one hand, capping the annual premium increase constrains insurers' ability to raise premiums and harvest inertial incumbent enrollees, which therefore tends to decrease price levels. On the other hand, given the reduced room for harvesting, insurers now face weaker incentives to set low prices to invest in future demand. The net impact on average price levels depends on the interactions of these two channels. Because there is strong competition among Part D sponsors, the invest incentive turns out to be less sensitive to this policy change than the harvest incentive, leading to a lower price level on average. One caveat with this prediction is that it might be specific to the market structure of Part D, and therefore it needs to be re-evaluated in other market settings.

Table 11 reports simulated per-period consumer surplus for the benchmark model as well as the three policy counter-factuals. For consumer surplus, I also show the difference between each counter-factual and the benchmark and decompose this difference into the component driven by changes in the share of standard enrollees choosing stand-alone prescription drug plans, the component driven by changes in prices and the component driven by changes in choice efficiency. Capping annual price increase results in the highest consumer welfare, which is largely due to the direct effect of lower average prices and the resulting increase in enrollment, but there is also a noticeable reduction in dynamic choice inefficiency. Offering a low-price public option increases consumer surplus, which works mostly through enrollment in the public option, but there is also a small reduction in dynamic choice inefficiency. Removing the risk corridors has little impact on all margins.

Table 12 reports results on social welfare, including consumer surplus, insurer profit, subsidy and social surplus. Capping premium increase is the most desirable in terms of both consumer and social welfare. Offering a low-price public option improves consumer welfare, but such welfare gains are dominated by the extra social cost of providing the public option. Removing the risk corridors has little impact on consumer and social welfare but transfers money from the government to insurers because with risk sharing, taxes on excessive gains outweigh subsidies on excessive losses both in the model and empirically.

These policy experiments are informative about the desirability of each policy in terms of restricting invest-then-harvest pricing and in terms of improving consumer welfare. Among the policies I consider, a policy change to cap premium increases would be the most effective in improving consumer welfare. This welfare increase comes from both smoother price dynamics and lower average premiums. There are two important next steps to check the robustness of the policy implications. First, I am working on models with a longer time horizon to evaluate the robustness of my conclusions. While key insights from these policy experiments are intuitive and the qualitative interpretations are not an artifact of the current two-period set up, some numbers might be different when we consider a longer time horizon. For example, prices in the second year in the model with inertia are higher than the counter-factual without inertia, which will change after allowing for a longer and more realistic time horizon. Second, the timing of policy intervention matters and I plan to evaluate effects of introducing the cap on annual premium increases at different points of time.

0.8 Conclusion

A growing literature has documented evidence that consumers in health insurance markets behave as if they face substantial switching costs when choosing health insurance plans. In this paper, I investigate whether private insurers in Medicare Part D exploit this type of consumer inertia when setting prices for insurance plans. I first document descriptive evidence consistent with insurers initially setting low prices in order to invest in future demand before later raising prices to harvest inertial consumers. To explore the implications of these invest and harvest incentives for equilibrium pricing, I develop and estimate a dynamic model of insurers' pricing decisions that incorporates demand inertia and adverse selection. I estimate a high discount factor among insurers, which is indicative of a strong incentive to invest in future demand and is consistent with low prices observed early on. I also find that on net, demand inertia reduces equilibrium prices, i.e. the invest incentive dominates the harvest incentive. Finally, I evaluate welfare consequences of policies that could be used to constrain insurers' ability to conduct such invest-then-harvest pricing patterns. Among the policies that I analyze, I find that a policy change to cap premium increases would be the most effective in improving consumer welfare by both lowering average premiums and smoothing prices over time.



Figure 0-1: Markup By Year of Entry

Notes: This figure shows trends in average annual markup for plans introduced in each year. The horizontal axis is year and the vertical axis is annual markup among standard enrollees. Each data point represents the enrollment-weighted average markup for a given cohort of plans in a given year. Each line shows the trend in average annual markup for plans introduced in a specific year.





(a) Expected claims cost by consumer type: low, medium, and high risk scores





Notes: Panel (a) reports, for each type of enrollee, their expected claims cost to a basic plan providing the required minimum coverage or standard defined benefit (SDB) and to an enhancedbenefit plan offering zero deductible and gap coverage on top of basic coverage. Standard enrollees are divided into low-, medium- and high-risk groups based on their risk scores. Panel (b) reports, for each type of enrollee, the expected claims cost as in Panel (a), switching cost in dollars, and willingness-to-pay for extra coverage (zero deductible and gap coverage) relative to basic coverage.

	Mean	SD	Min	Max			
Panel A: Stand-alone Prescription Drug Plans							
# Insurers	21	2.5	11	29			
# Plans	47	8.2	27	66			
# Basic Plans	24	4.0	15	36			
# Enhanced-Benefit Plans	23	4.9	12	31			
Herfindahl index	0.22	0.06	0.13	0.48			
Average Annual Premium, 2006	329	36.5	270	413			
Average Annual Premium, 2011	507	46.6	393	583			
Panel B: Standard Medicare Beneficiaries							
# Beneficiaries	217,794	160,446	8,635	762,538			
Annual Rival Rate	0.10	0.015	0.078	0.150			
Annual Attrition Rate	0.08	0.007	0.061	0.098			
Share in Stand-alone Plans, 2006	0.24	0.081	0.042	0.43			
Share in Stand-alone Plans, 2011	0.26	0.063	0.18	0.42			
Panel C: Low-income Medicare Beneficiaries							
# Beneficiaries	59,738	51,148	2,822	267,848			
Annual Rival Rate	0.13	0.022	0.092	0.25			
Annual Attrition Rate	0.10	0.012	0.073	0.15			
Share in Stand-alone Plans, 2006	0.88	0.098	0.52	1			
Share in Stand-alone Plans, 2011	0.80	0.11	0.46	0.99			

 Table 1: Summary Statistics

Notes: Panel A reports summary statistics for stand-alone plans across markets. The Herfindahl Index is computed using enrollment of standard enrollees. Average premiums are weighted by standard enrollment. Panels B and C report summary statistics on standard and low-income Medicare beneficiaries at the market level. Numbers of beneficiaries correspond to the 20% random sample and should be multiplied by 5 to get actual Medicare population size. Arrival and attrition rates are relative to lagged population size. Shares in stand-alone plans are calculated as out of the entire population of standard or low-income Medicare beneficiaries, including those with standalone plans, those with bundled coverage under Medicare Advantage, those with coverage provided by employers or third parties, and those without any prescription drug coverage.

	# Plans	# Entries	# Exits	Total Premium	Markup
Pooled	9490			371.9	47.5
2006	1429	1429	NA	329.0	-5.5
2007	1865	594	3	362.7	-55.4
2008	1824	201	86	408.1	4.2
2009	1687	53	83	476.6	71.9
2010	1576	107	24	503.3	148.8
2011	1109	22	95	507.1	121.3

Table 2: Summary Statistics on Plans by Year

Notes: The table reports summary statistics for all stand-alone prescription drug plans, first pooled across years and then by year. Total premium is the total of annual basic premium and annual supplemental premium. Premiums and markups are reported in enrollment-weighted averages. Markup is defined in Section 0.2.2.

Beneficiary Cohort	2006	2007	2008	2009	2010	2011
# Standard Beneficiaries in PDPs	1,412,073	126,234	121,569	111,783	109,327	118,387
Share in plans introduced in 06 (%)	83.68	79.14	76.79	71.90	71.67	71.67
Share in plans introduced in 07 (%)	7.15	8.10	9.22	10.42	8.42	7.06
Share in plans introduced in 08 (%)	2.60	2.79	3.09	3.97	3.32	2.03
Share in plans introduced in 09 (%)	1.22	1.78	2.16	3.23	2.34	0.97
Share in plans introduced in 10 (%)	3.13	4.76	5.15	6.30	8.85	9.44

3.43

3.58

4.18

2.23

8.83

5.41

Table 3: Enrollment Shares as of 2011 by Beneficiary Cohort

1

Share in plans introduced in 11 (%)

Notes: This table reports shares of standard enrollees choosing stand-alone plans by cohort of beneficiaries (the columns) and by year of plan entry (the rows). Beneficiary cohort is the year that the beneficiary first enrolls in a stand-alone prescription drug plan. Each cell reports the share of the corresponding enrollee cohort choosing plans introduced in a certain year. The enrollment shares in this table are computed as of 2011. For example, the first column shows that there were about 1.4 million standard Medicare beneficiaries who first enrolled in stand-alone prescription drug plans as standard enrollees in 2006. In 2011, among these standard beneficiaries, 83 percent were enrolled in plans introduced in 2006, 7 percent were enrolled in plans introduced in 2007, and so forth.

Markup	(1)	(2)	(3)	(4)
$1{Entry}, =1$ for entrants	-82.7	-182.8	-158.1	-147.7
	(64.9)	(42.8)	(30.8)	(27.9)
Market FE		x	x	x
Year FE		x	x	x
Insurer FE			x	x
Plan Features				x
N	9312	9312	9312	9312
Adjusted R^2	0.024	0.173	0.268	0.688

Table 4: Comparing Markups Between Entrant Plans and Incumbent Plans

Notes: The table reports the regression results for equation 1, using plan-year level observations for all plans in the sample period 2006-2011. The regressor of interest is a dummy variable equal to one if the plan enters in that year. All standard errors are clustered at the plan level. Column (1) reports estimates without any controls. Column (2) controls for market and year fixed effects. Column (3) also controls for insurer-fixed effects in addition to market and year fixed effects. Column (4) adds controls for plan coverage, including deductible amount, whether the plan offers gap coverage and tiered cost sharing. Standard errors are clustered at the insurer level.

	Experie	nced Prior to	(4) Experi-	(5)	
	(1) Major MA sponsors	(2) MA sponsors	(3) Medicare Sponsors	enced in Part D	Enhanced benefit plans
1 {Entry}, =1 for entrants	-186.6	-168.9	-168.6	-192.0	-134.3
	(29.6)	(25.1)	(24.9)	(30.5)	(62.1)
N	2881	5339	5657	2918	4573
Adjusted R^2	0.732	0.707	0.706	0.567	0.766

Table 5: Comparing Markups Between Entrant Plans and Incumbent Plans on Subsamples

Notes: The table reports the regression results for equation 1, using plan-year level observations for subsamples of plans in 2006-2011. The regressor of interest is a dummy variable equal to one if the plan enters in that year. Controls include plan coverage, and market, insurer and year fixed effects. Column (1) reports estimates using the subsample of plans offered by insurers that were major sponsors in Medicare Advantage prior to 2006 and offered prescription drug coverage bundled with medical coverage to the Medicare population. Column (2) reports estimates using the subsample of plans offered by insurers with some experience in Medicare Advantage prior to 2006. Column (3) reports estimates using the subsample of plans offered by insurers that provided insurance to the Medicare population prior to 2006. Column (4) reports estimates using the subsample of plans offered by insurers that have served at least 5000 Part D enrollees in the same market before. Column (5) reports estimates using the subsample of enhanced-benefit plans only, which are not eligible for random assignment of low-income beneficiaries. Standard errors are clustered at the insurer level.

	(1)	(2)	(3)	(4)
Annual Premium, \$100	-0.39	-0.45	-0.41	-0.50
	(0.01)	(0.01)	(0.01)	(0.01)
Default plan, $1/0$	5.45	5.61	5.07	5.09
	(0.25)	(0.26)	(0.26)	(0.26)
x Health Risk Score	0.23	0.22	0.36	0.37
	(0.06)	(0.07)	(0.07)	(0.07)
Number of insurer FE	3	3	10	10
Use lagged cost as IV for Premium	No	Yes	No	Yes
Implied SC for 75yo female, av. risk	\$1506	\$1330	\$1392	\$1164

Table 6: Polyakova (2015)'s Demand Estimates for Standard Enrollees

Notes: The table reports estimation results for a few key coefficients from Polyakova (2015)'s simulated maximum likelihood estimation – coefficients on plan premium, on the default dummy, and on the interaction of the default dummy and enrollee risk score. Columns 1 and 3 do not instrument for annual premium, while columns 2 and 4 use lagged cost as an instrument for premium. Columns 1 and 2 control for insurer dummies for the 2 biggest insurers in each market (the omitted category consists of all other insurers), while columns 3 and 4 control for insurer dummies for the 9 biggest insurers in each market (the omitted category consists of all other insurers).

	Single-plan	Multi-plan Insurers		
	Insurers			
-	Basic Plans	Basic Plans	Enhanced-Be	nefit Plans
	Bid for P_{basic}	Bid for P_{basic}	Bid for Pbasic	$P_{supplemental}$
Intercept	231.4	1065.5	1244.7	66.7
	(133.9)	(679.2)	(238.91)	(268.0)
Plan Coverage				
(%)				
Deductible (in \$100)			-56.1	-18.9
			(17.3)	(16.5)
1{Gap coverage}			146.5	95.5
			(29.3)	(18.0)
Lagged Shares By 7	Type of Standard	Enrollees (%)	、	· · ·
Low risk share	4.1	-4.2	-16.6	3.2
	(3.3)	(3.9)	(6.7)	(3.2)
Female low risk	-10.6	7.2	3.1	-1.1
	(3.9)	(6.1)	(4.0)	(2.5)
Male medium risk	3.2	0.8	-5.1	-0.5
	(5.9)	(2.7)	(10.9)	(5.7)
Female medium risk	-10.2	-11.8	-9.8	-16.9
	(5.3)	(4.7)	(7.3)	(6.9)
Male high risk	-1.2	-1.7	11.5	20.5
-	(8.7)	(4.2)	(8.8)	(6.7)
Female high risk	15.3	10.7	1.9	-5.5
-	(6.4)	(4.1)	(9.5)	(8.9)
Number of obs.	677	2804	3265	3278
Adjusted R^2	0.78	0.56	0.72	0.69

Table 7: Empirical Pricing Strategies

Notes: The table reports key coefficient estimates from empirical pricing policy function estimation for three clusters of plans, controlling for plan coverage and insurer fixed effects. All standard errors are clustered at the insurer level.

Table 8: Structural Parameter Estimate

	Coefficient	Standard Error
Discount Factor $(\hat{\beta})$	0.946	0.073

Notes: The table reports the minimum-distance estimate for the discount factor and the bootstrapped standard error.

Average Markup	Model w.	Restrict Insurers' Ability to Exploit Inertia				
	Inertia	Cap Annual	Add Public	Remove Risk		
		Price Rise	Option	Corridors		
2006	-8	157	45	7		
2007	403	164	346	398		

Table 9:	Decomposition	Results:	Equilibrium	Markup	Levels
----------	---------------	----------	-------------	--------	--------

Notes: The table reports enrollment-weighted average markups (among standard enrollees) in the two-period model with consumer inertia, in the counter-factual benchmark without inertia and in the counter-factual with myopic insurers.

 Table 10: Counter-factual Policy Experiments: Equilibrium Markup Levels

Average Markup	Model w.	Restrict Insurers' Ability to Exploit Inertia				
	Inertia	Cap Annual	Add Public	Remove Risk		
		Price Rise	Option	Corridors		
2006	-8	157	45	7		
2007	403	164	346	398		

Notes: The table reports enrollment-weighted average markups (among standard enrollees) in the two-period model with consumer inertia and in the counter-factual policy simulations.

Consumer Surplus (\$Millions)	Model w.	Restrict Insurers' Ability to Exploit Inertia			
Consumer Sarpius (ummons)	Inertia	Cap Annual	Add Public	Remove	
		Price Rise	Option	Risk	
				Corridors	
Consumer Surplus	78.68	109.23	93.24	78.88	
ΔCS		30.54	14.56	0.20	
ΔCS due to ΔP		11.30	0.48	-1.42	
ΔCS due to $\Delta choice$ efficiency		6.87	3.60	1.52	
ΔCS due to ΔPDP share		12.37	10.48	0.10	

Table 11: Counter-factual Policy Experiments: Consumer Welfare

Notes: The table reports consumer welfare estimates for the actual policy setting (as the benchmark for comparison) and for policy counter-factuals (relative to the benchmark), all with a simplified two-period setting. Consumer Surplus is as defined in Section 5.3 and is converted to a per-year value so that it can be consistently compared across counter-factuals. The change in consumer surplus resulting from each counter-factual policy is decomposed into three components – the difference due to enrollees opting in and out of stand-alone plans, the difference due to the resulting equilibrium price changes, and the difference due to changes in consumers' plan choices.

Social Wolfers (Millions)	Model w.	Restrict Insu	estrict Insurers' Ability to Exploit Inertia			
Social Wenare (aminons)	Inertia	Cap Annual	Cap Annual Add Public			
		Price Rise	Option	Risk		
				Corridors		
Consumer Surplus	78.68	109.23	93.24	78.88		
Insurer Profit	14.43	16.01	9.98	46.60		
Direct Subsidy	182.81	165.64	157.06	183.60		
Subsidy for Public Option	NA	NA	13.75	NA		
Total Direct Subsidy, rt MA-PD	2.88	-19.01	5.82	2.88		
Risk Corridor Payments to insurers	-30.34	-27.17	-31.30	0.00		
Total Surplus	119.71	177.12	109.09	121.74		
Public Option Share	NA	NA	0.08	NA		

 Table 12: Counter-factual Policy Experiments: Social Welfare

Notes: The table reports welfare estimates for the actual policy setting (as the benchmark for comparison) and for policy counter-factuals, all with a simplified two-period setting. Consumer Surplus, Insurer Profit, Direct Subsidy and Risk Corridor Payments are as defined in Section 5.3 and are all converted to a per-year value so that they can be consistently compared across counter-factuals. Positive risk corridor payments mean that the government pays insurers in aggregate while negative risk corridor payments mean that insurers pay the government in aggregate.

0.8.1 Appendix



Figure A.1: Expected claims cost, switching cost and preference by consumer type

Notes: This figure reports, for each type of enrollees, their expected cost, switching cost in dollars, and willingness-to-pay for extra coverage: zero deductible and gap coverage, relative to basic coverage. In addition to the division into low-, medium- and high-risk groups based on their risk scores as in Figure 0-2, standard enrollees are also grouped by gender, which makes a small difference in terms of expected cost.

	Low Risk-score		Medium	Medium Risk-score		sk-score	Low-income
	Male	Female	Male	Female	Male	Female	Enrollees
Intercept	383.4	383.2	750.2	706.9	1145.2	1078.6	1018.4
	(5.11)	(4.44)	(4.91)	(3.50)	(9.04)	(4.85)	(3.35)
Deductible	-0.585	-0.451	-0.297	-0.234	0	0.056	0.248
	(0.015)	(0.013)	(0.014)	(-0.010)	(0.02)	(0.013)	(0.009)
Partial Gap Cov.	206.2	204.5	192.2	226.8	266.4	344.1	594.2
	(9.3)	(7.8)	(6.5)	(5.0)	(10.0)	(6.0)	(14.2)
Full Gap Cov.	637.3	653.9	716.8	694.3	1051.8	1032.8	1490.0
	(19.4)	(15.2)	(12.1)	(8.8)	(15.6)	(8.9)	(26.4)
Number of obs.	568,117	670,788	1,195,847	1,856,589	1,160,410	$2,\!451,\!539$	9,928,135
Adjusted \mathbb{R}^2	0.08	0.09	0.07	0.08	0.05	0.07	0.02

Table A.1: Individual Cost Estimation

Notes: The table reports some key coefficients from the individual cost estimation for Equation 5. Insurer fixed effects are included. Standard errors are clustered at the insurer level.

Bibliography

- Abaluck, J., Gruber, J., June 2011. Choice inconsistencies among the elderly: Evidence from plan choice in the medicare part D program. The American Economic Review 101, 1180–1210.
- Abaluck, J., Gruber, J., June 2013. Evolving choice inconsistencies in choice of prescription drug insurance. Working Paper 19163, National Bureau of Economic Research.
- Arie, G., Grieco, P. L., 2014. Who pays for switching costs? Quantitative Marketing and Economics 12 (4), 379–419.
- Bajari, P., Benkard, C. L., Levin, J., 2007. Estimating dynamic models of imperfect competition. Econometrica 75 (5), 1331–1370.
- Bajari, P., Hong, H., Ryan, S. P., 2010. Identification and estimation of discrete game of complete information. Econometrica 78 (5), 1529–1568.
- Beggs, A., Klemperer, P., 1992. Multi-period competition with switching costs. Econometrica 60 (3), 651–666.
- Benkard, C. L., 2004. A dynamic analysis of the market for widebodied commercial aircraft. Review of Economic Studies 71, 581–611.
- Boards of Trustees, 2012. 2012 annual report of the boards of trustees of the federal hospital insurance and federal supplementary insurance trust funds.
- Decarolis, F., 2015. Medicare part D: Are insurers gaming the low income subsidy design? American Economic Review 105 (4), 1547–80.
- Decarolis, F., Polyakova, M., Ryan, S. P., June 2015. The welfare effects of supply-side regulations in medicare part d. Working Paper 21298, National Bureau of Economic Research.
- Dubé, J.-P., Hitsch, G. J., Rossi, P. E., 2009. Do switching costs make markets less competitive? Journal of Marketing Research 46 (4), 435–445.
- Einav, L., Finkelstein, A., Schrimpf, P., forthcoming. The response of drug expenditures to nonlinear contract design: Evidence from medicare part d. The Quarterly Journal of Economics.

- Ericson, K. M., 2014. Consumer inertia and firm pricing in the medicare part D prescription drug insurance exchange. American Economic Journal: Economic Policy 6 (1), 38–64.
- Ericson, K. M., Starc, A., 2015. Pricing regulation and imperfect competition on the massachusetts health insurance exchange. Review of Economics and Statistics 97 (3), 667–682.
- Farrell, J., Klemperer, P., 2007. Coordination and lock-in: Competition with switching costs and network eects. In: Armstrong, M., Porter, R. (Eds.), Handbook of Industrial Organization. Vol. 3. Elsevier B.V., pp. 1967–2072.
- Gold, M., March 2006. The landscape of private firms offering medicare prescription drug coverage in 2006.
- Handel, B. R., 2014. Adverse selection and inertia in health insurance markets: When nudging hurts. American Economic Review 103 (7), 2643–82.
- Hausman, J. A., Poterba, J. M., Summer 1987. Household Behavior and the Tax Reform Act of 1986. Journal of Economic Perspectives 1 (1), 101–19.
- Heiss, F., Leive, A., Mcfadden, D., Winter, J., 2013. Plan selection in medicare part D: Evidence from administrative data. Journal of Health Economics 32 (6), 1325–1344.
- Heiss, F., McFadden, D., Winter, J., 2008. Mind the gap! consumer perceptions and choices. Tech. rep., Munich Center for the Economics of Aging (MEA) at the Max Planck Institute for Social Law and Social Policy.
- Ho, K., Hogan, J., Scott Morton, F., March 2015. The Impact of Consumer Inattention on Insurer Pricing in the Medicare Part D Program. Working Paper 21028, National Bureau of Economic Research.
- Hoadley, J., Cubanski, J., Summer, L., Huang, J., October 2012. Medicare policy data spotlight medicare part D: A first look at part D plan offerings in 2013Kaiser Family Foundation Report.
- Hoadley, J., Summer, L., Hargrave, E., Cubanski, J., Neuman, T., August 2014. Medicare part D in its ninth year: The 2014 marketplace and key trends, 2006–2014Kaiser Family Foundation Report.
- Kesternich, I., Heiss, F., Mcfadden, D., Winter, J., 2013. Suit the action to the word, the word to the action: Hypothetical choices and real decisions in medicare part D. Journal of Health Economics 32 (6), 1313–1324.
- Ketcham, J. D., Lucarelli, C., Miravete, E. J., Roebuck, M. C., 2012. Sinking, swimming, or learning to swim in medicare part D. American Economic Review 102 (6), 2639–73.
- Ketcham, J. D., Lucarelli, C., Powers, C. A., 2015. Paying attention or paying too much in medicare part d. American Economic Review 105 (1), 204–33.

- Klemperer, P., 1987. Markets with consumer switching costs. The Quarterly Journal of Economics 102 (2), 375–94.
- Kling, J. R., Mullainathan, S., Shafir, E., Vermeulen, L. C., Wrobel, M. V., 2012. Comparison friction experimental evidence from medicare drug plans. The Quarterly Journal of Economics 127 (1), 199–235.
- Knittel, C., 1997. Interstate long distance rates: Search costs, switching costs, and market power. Review of Industrial Organization 12 (4), 519–536.
- Krasner, J., December 31 2006. Insurer hits millions of seniors with drug cost hike. Retrieved from http://www.boston.com.
- Larkin, I., 2008. Bargains-then-ripoffs: Innovation, pricing and lock-in in enterprise software. Academy of Management Proceedings.
- Miller, D. P., 2015. Risk adjustment and low income subsidy distortions in medicare part DMimeo, Clemson University.
- Miller, D. P., Yeo, J., 2015a. Discrete Choice Models to Estimate Switching Costs: An Application to Medicare Part DMimeo, Clemson University and Singapore Management University.
- Miller, D. P., Yeo, J., 2015b. The Consequences of a Public Health Insurance Option: Evidence From Medicare Part DMimeo, Clemson University and Singapore Management University.
- Miller, K., December, 9 2014. Do private medicare firms have lower costs? Job market paper, Department of Economics, University of Minnesota.
- Nosal, K., 2012. Estimating switching costs for medicare advantage plansMimeo, University of Mannheim.
- Park, M., 2010. The economic impact of wireless number portabilityWorking Paper.
- Polyakova, M., September 2015. Regulation of insurance with adverse selection and switching costs: Evidence from medicare part D. Working Paper 21541, National Bureau of Economic Research.
- Sharpe, S. A., 1997. The effect of consumer switching costs on prices: A theory and its application to the bank deposit market. Review of Industrial Organization 12, 79–94.
- Shepard, M., 2015. Hospital Network Competition and Adverse Selection: Evidence from the Massachusetts Health Insurance Exchange. Job market paper, Harvard University.
- Shi, M., Chiang, J., Rhee, B., 2006. Price competition with reduced consumer switching costs: The case of "wireless number portability" in the cellular phone industry. Management Science 52 (1), 27–38.

- Stango, V., 2002. Pricing with consumer switching costs: Evidence from the credit card market. The Journal of Industrial Economics 50 (4), 475–492.
- Starc, A., 2014. Insurer pricing and consumer welfare: evidence from medigap. The RAND Journal of Economics 45 (1), 198–220.
- Summer, L., Hoadley, J., Hargrave, E., September 2010. The medicare part d low-income subsidy program: Experience to date and policy issues for considerationKFF Report.
- U.S. Department of Health and Human Services, September 2014. U.S. Department of Health and Human Services Rate Review Annual Reporthttp://aspe.hhs.gov/health/reports/2014/RateReview/rpt_RateReview.pdf.
- Viard, V. B., 2007. Do switching costs make markets more or less competitive? the case of 800number portability. The RAND Journal of Economics 38 (1), 146–163.
- Waterson, M., 2003. The role of consumers in competition and competition policy. International Journal of Industrial Organization 21 (2), 129–150.
- Weintraub, G. Y., Benkard, C. L., Van Roy, B., 2008. Markov perfect industry dynamics with many firms. Econometrica 76 (6), 1375–1411.